Original	Sheet No. TITLE	RECEN
Cancelling	Sheet No.	JAN 19 STATE OF ALA
		REGULATORY COMMISSIO
Storage Alaska, LLC		
	D 1	
	5. 1	
OF		
0F		
ILET NATURAL GAS S	TORAGE ALASKA LLO	
LET NATURAL GAS S	TORAGE ALASKA, ELG	
0 Spenard Road, Ancho	orage Alaska 99503	
latural Gas Storage Serv	vice to Southcentral Alask	a
9)	Effective: January 31, 20	011
	Storage Alaska, LLC TARIFF NO OF LET NATURAL GAS S 0 Spenard Road, Ancho atural Gas Storage Ser	Cancelling Sheet No. Storage Alaska, LLC TARIFF NO. 1 OF LET NATURAL GAS STORAGE ALASKA, LLC 0 Spenard Road, Anchorage Alaska 99503 atural Gas Storage Service to Southcentral Alask

M. Colleen Starring

((; ;

(..... (**...*

÷. .

CINGSA Proposed Tariff filed January 19, 2011 U-10-51(9) Appendix D Page 1 of 77 Seventh Revision Cancelling Sixth Revision

RCA No. 733

Sheet No.

1

1

Sheet No.

FEB 22 2021 STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

RECEIVED

Cook Inlet Natural Gas Storage Alaska, LLC

	JTHORIZED SERVICE AREA D REGULATIONS	
Section 1	Territory	
Section 2	Definitions	
Section 3	Storage Service Agreements	
Section 3	Firm Storage Service (FSS) Terms and Conditions	
Section 5	Interruptible Storage Service (ISS) Terms and Conditions	
Section 6	Term	
Section 7	Expiration of Term	
Section 8	Facilities for Additional Services	
Section 9	Point(s) of Injection/Withdrawal	
Section 10	Quality of Gas	
Section 11	Measurement and Measurement Equipment	
Section 12	Force Majeure and Remedies	
Section 13	Delivery Pressure	
Section 14	Billing and Payment	
Section 15	Default Suspension and Termination	
Section 16	Non-Waiver of Future Defaults	
Section 17	Requests for Storage Service	
Section 18	Nominations, Scheduling and Allocations	
Section 19	Operational Information	
Section 20	Right to Commingle	
Section 21	Title Transfer of Gas in Storage	
Section 22	Responsibility for Gas	
Section 23	Warranty of Title	
Section 24	Limitation of Liability	
Section 25	Responsibility for Associated Transportation	
Section 26	Creditworthiness	
Section 27	Regulations	
Section 28	Transfer and Assignment	
Section 29	CINGSA Facility Expansion and Facility Capacity Rights	
Section 30	Fuel Use Percentage	
Section 31	Operational Purchases and Sales of Gas	
Section 32	Operational Flow Order ("OFO")	70
Section 33	Right of First Refusal (ROFR)	
Section 34	General Terms and Conditions	
Section 35	Rate Schedules and Other Charges	
Section 36	Schedule of Special Contracts	82
Section 37	Crediting of Non-Firm Revenues to FSS Customers	83
Section 38	Well Maintenance Surcharge Mechanism	
Section 39	Customer Compliant Procedures	
Section 40	Formula Rate Mechanism	

Ν





Pursuant to U-19-025(20)

Effective: August 5, 2020

Issued By: Cook Inlet Natural Gas Storage Alaska, LLC

RCA No. 733	Third Revision	Sheet No.	3	RECEIVED
10/110:700	Cancelling	oncer no.	Ŭ	January 31, 2020
	Second Revision	Sheet No.	3	STATE OF ALASKA
				REGULATORY COMMISSION OF ALASKA

RULES AND REGULATIONS

This tariff contains the rules and rates of Cook Inlet Natural Gas Storage Alaska, LLC, from here forward called the "Seller" or the "Company". Seller receives, injects, stores, withdraws and delivers natural Gas transported in intrastate commerce subject to jurisdiction of the Regulatory Commission of Alaska.

SECTION 1 TERRITORY

1.1 Applicability

These rules and regulations shall be applicable to all Customers. The service area map is located on Sheet 2. The service area is more fully described as:

Geographic Area:

T5N R11W Sections 3 through 10 and 15 through 18 (All the above with reference to the Seward Meridian) Facilities:

- 1. the Sterling C Gas Storage Pool of the Cannery Loop Unit, consisting of the interval that is common to, and correlating with, the measured depths from 6,690 ft. to 6,945 ft. in well CLU No.8;
- 2. the wells required to access the Sterling C Gas Storage Pool; and
- 3. all associated surface facilities.

Pressure:

The Sterling C Gas Storage Pool may be developed and operated as a gas storage field with a maximum allowable reservoir pressure of 2,200 Psia.

1.2 Office and Tariff Location

A copy of this tariff is on file for inspection upon request by any member of the general public during regular business hours (9 AM to 4 PM ACT each Business Day). The office is located at 3000 Spenard Road, Anchorage, Alaska.

A complete copy of this tariff can also be found on the Company's website at: www.cingsa.com

т

Ν

Ν

RCA No. 73	3 Third Revision Sheet No. 4 RECEN Cancelling January 31 Second Revision Sheet No. 4 state of al regulatory commis				
	t Natural Gas Storage Alaska, LLC				
SECTION 2	DEFINITIONS				
2.1	The term "ADNR" shall mean the Alaska Department of Natural Resources.	D T			
2.2	The terms "Agreement", "Service Agreement" and "Gas Storage Agreement" shall mean either the FSS Agreement executed pursuant to Section 4 of these Rules and Regulations by the Customer and Seller and any exhibits, attachments and/or amendments thereto, or the ISS Agreement executed pursuant to Section 5 of these Rules and Regulations by the Customer and Seller and any exhibits, attachments and/or amendments and/or amendments thereto.	T,C			
2.3	The term "Alaska Clock Time" or "ACT" shall mean Alaska Daylight Time when Daylight Savings Time is in effect and Alaska Standard Time when Daylight Savings Time is not in effect.	т			
2.4	The term "APC/ENSTAR Line" shall mean the Alaska Pipeline Company and ENSTAR Natural Gas Company pipeline system that interconnects with the CINGSA Facility.	N N N			
2.5	The term "Allocated Expansion Capacity" shall have the meaning ascribed to such term in Section 29.1c).				
2.6	The term "Base Storage Design" shall have the meaning described in the definition of the "CINGSA Facility" in Section 2.10.	т			
2.7	The term "Bcf" shall mean one billion (1,000,000,000) Standard Cubic Feet.				
2.8	The term "Business Day" shall mean every Monday, Tuesday, Wednesday, Thursday, or Friday, excluding all Company holidays.	C C			

D – Section 2.1 defining "Adjusted Rate Schedule FSS" published on the Second Revision of Sheet 4 has been deleted.

				RECEIVED
RCA No. 733	Fourth Revision Cancelling	Sheet No.	5	NOV 03 2023
	Third Revision	Sheet No.	5	STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

Cook Inlet Natural Gas Storage Alaska, LLC

- 2.9 The term "Btu" ("British thermal unit") shall mean the amount of heat required to raise the temperature of one (1) pound of water one degree Fahrenheit (1°F) at sixty degrees Fahrenheit (60°F). (Btu is measured on a dry basis at 14.65 pounds per square inch absolute (Psia).)
 - The terms "CINGSA Facility", "Storage Facility", "Storage Facilities" and 2.10 "Storage Project" shall mean the underground natural Gas storage facilities located and developed by Cook Inlet Natural Gas Storage Alaska, LLC (Seller) Т in the Kenai Peninsula Borough of Alaska, with an initial "Base Storage Design" consisting of (1) a design working storage capacity of 11 Bcf, (2) a design maximum daily withdrawal quantity of 150,000 Mcf/d, and (3) a design maximum daily injection quantity of 150,000 Mcf/d, together with and including any and all related above or below ground and other real and/or personal property, tangible and/or intangible, that was or is to be developed, acquired, owned and/or held by Seller for the purpose of commercializing same as a natural Gas storage services provider, including sufficient quantities of base Gas, either already existing in place or acquired from third parties and provided С by Seller. The 2023 Expansion Project adds (1) a design working storage capacity of 2 Bcf, (2) a design maximum daily withdrawal quantity of 65,000 С С Mcf/d, and (3) a design maximum daily injection quantity of 75,000 Mcf/d for a С total storage design of (1) a design working storage capacity of 13 Bcf, (2) a С design maximum daily withdrawal quantity of 215,000 Mcf/d, and (3) a design С maximum daily injection quantity of 225,000 Mcf/d.
 - 2.11 The term "Commencement Date" shall mean three (3) Business Days following a notice and representation from Seller to Customer that the CINGSA Facility is commercially available to provide firm injection service to Customer under the FSS Agreement. The appropriate notice was provided on March 27, 2012 and C the CINGSA Facility began injections of gas on April 1, 2012.
 - 2.12 The terms "Company" and "Storage Provider" shall mean Cook Inlet Natural Gas Storage Alaska, LLC, (CINGSA) its successors and assigns.
 - 2.13 The term "Contract Injection Quantity" or "CIQ" shall mean maximum daily rate in Mcf/d of Gas which the Customer may inject into the CINGSA Facility on any given Day during each Contract Year, as provided for in Customer's Agreement.
 - 2.14 The term "Contract Withdrawal Quantity" or "CWQ" shall mean the maximum daily rate in Mcf/d of Gas which the Customer may withdraw from the CINGSA Facility on any given Day during each Contract Year, as provided for in Customer's Agreement.

				RECEIVED
RCA No. 733	Fourth Revision Cancelling	Sheet No.	6	NOV 03 2023
	Third Revision	Sheet No.	6	STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

55-733	Effective: December 18, 2023	
2.24	The term "Excess Storage Service" shall have the meaning ascribed to such term in Section 4.8.	
2.23	The term "Equivalent Quantities" shall mean a quantity of Gas containing an amount of Mcfs equal to the amount of Mcfs received by Seller for the account of Customer at the Point of Injection/Withdrawal reduced by the Mcfs removed for Seller's injection and/or withdrawal as attributable to the storage of Customer's Gas.	
2.22	The term "Early Termination Date" shall have the meaning ascribed to such term in Section 15.2a).	
2.21	The term "Dekatherm" (or "Dth") shall mean the quantity of heat energy which is equivalent to one (1) million (1,000,000) BTU and "MMDth" shall mean one million Dekatherms.	
2.20	The term "Day" shall mean a period of twenty-four consecutive hours, beginning at 12:00 a.m. ACT and ending on the following 11:59:59 p.m. ACT.	
2.19	The terms "Customer" and "Storage Customer" shall mean any individual, partnership, association, public or private corporation, or governmental agency being served with or applying for Gas storage service from Company at the Point(s) of Injection/Withdrawal under and pursuant to this Tariff. The term "Initial Capacity Customer" shall have the meaning ascribed to such term in Section 4.1d). The term "Expansion Customer" shall mean any Customer being served under an Expansion FSS Agreement.	C C C C C
2.18	The term "Current Fuel Use Surcharge Percentage" shall have the meaning ascribed to such term in Section 30.4.	
2.17	The term "Current Injection/Withdrawal Fuel Use Percentage" shall have the meaning ascribed to such term in Section 30.3.	
2.16	The term "Credit Alternative" shall have the meaning ascribed to such term in Section 26.3.	
2.15	The term "Contract Year" shall mean the period from April 1 of a calendar year through March 31 of the following calendar year or such shorter period as Customer and Seller may agree to.	

RCA No. 733	Third Revision	Sheet No.	7	RECEIVED
	Cancelling		•	January 31, 2020
	Second Revision	Sheet No.	7	STATE OF ALASKA
				REGULATORY COMMISSION OF ALASKA

2.25	The term "Expansion FSS Agreement" shall have the meaning ascribed to such term in Section 29.1d).	Т
2.26	The term "Expansion Precedent Agreement" shall have the meaning ascribed to such term in Section 29.1d).	т
2.27	The term "Expansion Project" shall have the meaning ascribed to such term in Section 29.1c).	т
2.28	The terms "Firm Storage Service" and "FSS" shall mean firm Gas storage services pursuant to Section 4 and Rate Schedule FSS.	T T
2.29	The term "Force Majeure Event" shall have the meaning ascribed to such term in Section 12.1.	т
2.30	The term "FSS Agreement" shall mean the Gas Storage Agreement for FSS executed pursuant to Section 4 of these Rules and Regulations by the Customer and Seller and any exhibits, attachments and/or amendments thereto.	т
2.31	The term "FSS Capacity Rate" shall have the meaning ascribed to such term in Section 4.6b).	T T
2.32	The term "FSS Injection/Withdrawal Commodity Rate" shall have the meaning ascribed to such term in Section 4.6e).	T T
2.33	The term "FSS Reservation Rate" shall have the meaning ascribed to such term in Section 4.6a).	т
2.34	The term "Fuel Use" shall have the meaning ascribed to such term in Section 30.1.	T T
2.35	The term "Gas" shall mean natural gas as produced in its natural state, natural gas that has been previously liquefied and restored to its gaseous state prior to delivery, gas synthesized or manufactured from oil, naphtha, coal or any other material that meets the quality standards contained in these Rules and Regulations and which Customer or Seller may elect to deliver and redeliver in lieu of or commingled with one or more of the types of gas described herein.	т

RCA No. 733	Fourth Revision	Sheet No.	8	RECEIVED
	Cancelling Third Revision	Sheet No.	8	January 31, 2020 STATE OF ALASKA REGULATORY COMMISSION OF ALASKA
Cook Inlet N	atural Gas Storage	Alaska, LLC		
				D

			D	
2	2.36	The term "Gas Flow Computer" or "GFC" shall have the meaning ascribed to such term in Section 11.2.	D	
2	2.37	The term "Guarantor" shall have the meaning ascribed to such term in Section 26.3a).		
2	2.38	The term "Injection/Withdrawal Fuel Use Percentage" shall have the meaning ascribed to in Section 30.5 and is set out in Section 35.3 a).	т	
2	2.39	The terms "Interruptible Storage Service" and "ISS" shall mean all interruptible storage-related services pursuant to Section 5 and Rate Schedule ISS.	т	
2	2.40	The term "ISS Agreement" shall mean the Gas Storage Agreement for ISS executed pursuant to Section 5 of these Rules and Regulations by the Customer and Seller and any exhibits, attachments and/or amendments thereto.		
2	2.41	The term "ISS Injection/Withdrawal Commodity Rate" shall have the meaning ascribed to such term in Section 5.3 b).		
2	2.42	The term "KBPL" shall mean the Kenai Beluga Pipeline which interconnects with the CINGSA Facility.	С	
2	2.43	The term "Lost and Unaccounted For" shall have the meaning ascribed to such term in Section 30.1.	т	
2	2.44	The terms "Maximum Daily Injection Quantity" and "MDIQ" shall mean maximum daily rate, in Mcf/d of Gas, which the Customer may inject on a given Day, as adjusted for the Storage Quantity, as provided for in Customer's Agreement.		
		n 2.35 previously published on the Third Revision of Sheet 8 has been deleted definition has been combined with the definition in Section 2.2.		

RCA No. 733	Fourth Revision	Sheet No.	9	RECEIVED
	Cancelling Third Revision	Sheet No.	9	January 31, 2020 STATE OF ALASKA
				REGULATORY COMMISSION OF ALASKA

	2.45	The terms "Maximum Daily Withdrawal Quantity" and "MDWQ" shall mean maximum daily rate, in Mcf/d of Gas, which the Customer may withdraw on a given Day, as adjusted for the Storage Quantity, as provided for in Customer's Agreement.	
	2.46	The terms "Maximum Storage Quantity" and "MSQ" shall mean maximum quantity, in Mcf of Gas, which the Customer may store on a given Day, as provided for in Customer's Agreement.	
	2.47	The term "Mcf" shall mean one thousand (1,000) Standard Cubic Feet and the term "Mcf/d" shall mean Mcf per Day.	
	2.48	The term "Month" shall mean the period beginning at midnight ACT on the first Day of a calendar month and ending at 11:59:59 p.m. ACT on the last Day of the calendar month.	
	2.49	The term "Net Present Value" or "NPV" shall mean the present value of future expected payment obligations for a term of up to twenty (20) years, using a discount rate of the Prime rate plus 2 percent.	
	2.50	The term "Notice of Request" shall have the meaning ascribed to such term in Section 29.1a).	
	2.51	The term "Operational Balancing Agreement" or "OBA" shall mean a contract between two parties which specifies the procedures to manage operating variances at an interconnection point.	
	2.52	The terms "Operational Flow Order" and "OFO" shall mean an order issued to alleviate conditions which threaten or could threaten the safe operations or system integrity of Seller's system or to maintain operations required to provide efficient and reliable Firm Storage Service. Whenever Seller experiences these conditions, any pertinent order will be referred to as an OFO.	т
1			

RCA No. 733	Third Revision	Sheet No.	10	RECEIVED
	Cancelling		-	January 31, 2020
	Second Revision	Sheet No.	10	STATE OF ALASKA

The term "Operator" shall have the meaning ascribed to such term in Section 2.53 11.3. 2.54 The term "Overrun Storage Services" shall have the meaning ascribed to such term in Section 4.7. 2.55 The term "Parties" shall mean Seller and Customer. 2.56 The term "Party" shall mean either Seller or Customer. 2.57 The term "Point(s) of Injection/Withdrawal" shall mean any or all points of C interconnection between Seller's CINGSA Facility and a Transporter's pipeline С as delineated in Section 9.1. The Point(s) of Injection/Withdrawal is the point of C measurement. 2.58 The term "Prior Period Adjustments" or "PPA" shall have the meaning ascribed Т to such term in Section 11.7. 2.59 The term "Prospective Expansion Customer(s)" shall have the meaning ascribed to such term in Section 29.1c). 2.60 The term "Psia" shall mean pounds per square inch absolute. 2.61 The term "Psig" shall mean pounds per square inch gauge. 2.62 The term "Ratchets" shall mean the reductions to MDWQ and MDIQ, as applicable, below the CIQ and CWQ based on the Customer's Storage Quantity as a percentage of MSQ, as stated in the FSS Agreement. Reserved for future use. 2.63 2.64 The term "RCA" or "Commission" shall mean the Regulatory Commission of **T** Alaska or any state commission, agency or other governmental body or bodies succeeding to the powers of such Commission.

REGULATORY COMMISSION OF ALASKA

RCA No. 733	Fourth Revision	Sheet No.	11	RECEIVED
	Cancelling			January 31, 2020
	Third Revision	Sheet No.	11	STATE OF ALASKA
				REGULATORY COMMISSION OF ALASKA

	t Matural Ous Otorage Alaska, LLO	
2.65	The term "Requested Expansion Capacity" shall have the meaning ascribed to such term in Section 29.1a).	
2.66	The term "Seller" shall mean Cook Inlet Natural Gas Storage Alaska, LLC.	
2.67	The term "Seller's Injection/Withdrawal Fuel Use" shall have the meaning ascribed to such term in Section 30.2.	
2.68	The term "Service Day" shall mean the Day during which Customer receives Storage Service pursuant to the nomination in accordance with Section 18.	
2.69	The term "Storage Quantity" shall mean the quantity of Gas held in storage at any time, by Seller, for the account of the Customer.	T T
2.70	The term "Standard Cubic Foot" shall mean the amount of Gas that would occupy a volume of one cubic foot at a temperature of sixty degrees (60°) Fahrenheit and at a pressure of fourteen and sixty-five-hundredths (14.65) Psia.	
2.71	The term "Tangible Net Worth" shall have the meaning ascribed to such term in Section 26.1c).	
2.72	The term "Tariff," or any other reference to the Tariff's "Rates," "Rate Schedule(s)," and/or "Rules and Regulations" shall mean, as the context requires, the rates, rate schedules, rules, regulations, terms and conditions herein.	
	(The next Sheet is Sheet 11.1)	

RCA No. 733	Second Revision	Sheet No.	11.1	RECEIVED
	Cancelling			January 31, 2020
	First Revision	Sheet No.	11.1	STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

- 2.73 The term "Total Contract Quantity" shall be the quantity that will be billed at the FSS Capacity Rate each month, as provided in Section 4.6 a) 1), and as stated T in the Agreement.
 - 2.74 The term "Total Heating Value per Cubic Foot" shall mean the number of Btu's produced by the combustion, at constant pressure, of one (1) dry cubic foot of Gas at an absolute pressure of fourteen and sixty five hundredths pounds (14.65#) per square inch and at a temperature of sixty degrees Fahrenheit (60°F), with air of the same pressure and temperature as the Gas, when the products of combustion are cooled to the initial temperature of the Gas and air, and when the water formed by combustion is condensed to the liquid state.
 - 2.75 The term "Transporter(s)" shall mean either Kenai Beluga Pipeline LLC, the owner and operator of the KBPL that interconnects with the CINGSA Facility or Alaska Pipeline Company and ENSTAR Natural Gas Company, the owner and operator of the APC/ENSTAR Line that also interconnects with the CINGSA Facility, or both.

(The next Sheet is Sheet 12)

RCA No. 733	First Revision	Sheet No.	12	RECEIVED
100/110.700	Cancelling	Oncer No.	12	January 31, 2020
	Original	Sheet No.	12	STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

SECTION 3 STORAGE SERVICE AGREEMENTS

An Agreement for storage service shall be entered into between Seller and Customer. In the event of conflict between these Rules and Regulations and the Agreement, the terms and conditions of the Agreement shall control.

Т

			40	RECEIVED
RCA No. 733	Second Revision Cancelling	Sheet No.	13	NOV 03 2023
	First Revision	Sheet No.	13	STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

SECTION 4	FIRM STORAGE SERVICE (FSS) TERMS AND CONDITIONS	
4.1	Availability: Commencing on the Commencement Date, Firm Storage Service (FSS) is available to any Customer qualifying for natural gas storage service if:	
	 Seller has determined that it has sufficient operationally available and uncommitted firm storage capacity and injection and withdrawal capacity to perform the service requested by Customer; and 	
	b) Seller and Customer have executed an Agreement under this Section 4; and	
	c) Customer accepts responsibility for arranging any transportation service required for utilization of the storage service provided under these Rules and Regulations and the Rate Schedule FSS.	
	d) In the case of Customers who enter into FSS Agreements before the Commencement Date ("Initial Capacity Customer"), the conditions stated above in Section 4.1a) through 4.1c) are deemed to have been met.	т
4.2	Except in the case of Customers who enter into FSS Agreements before the Commencement Date, availability of service under these Rules and Regulations and Rate Schedule FSS shall be subject to a determination by CINGSA that its performance of the service requested hereunder will not cause a reduction in CINGSA's ability to provide FSS under currently effective FSS Agreements;	

RCA No. 733	Third Revision	Sheet No.	14	RECEIVED
	Cancelling Second Revision	Sheet No.	14	April 13, 2020
	Decond Revision	oneet no.	14	STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

- 4.3 CINGSA is not required to construct, modify, or acquire any facilities to provide T FSS for a Customer who requests FSS under these Rules and Regulations T and Rate Schedules after the Commencement Date. This Section does not T apply to Requested Expansion Capacity or an Expansion Project under T Section 29.
 - 4.4 Seller shall have the right to curtail on a pro rata basis or discontinue FSS, in whole or in part, on all or a portion of the CINGSA Facility at any time for reasons of Force Majeure or to make scheduled modifications, repairs, operating changes, or for scheduled maintenance or testing, subject to Section 34.2 hereof. Seller shall issue an OFO (as defined in Section 32) when it identifies an occurrence of any condition that is likely to lead to a curtailment of an FSS Customer's firm deliverability.
 - 4.5 Applicability and Characteristics of Service: Section 4 shall apply to all FSS which is rendered by Seller for Customer pursuant to an executed Agreement under this Section. Storage Service under this Section shall be firm up to the MSQ and shall be firm up to the MDWQ and MDIQ on any Day. The MSQ, the CIQ, the CWQ, the MDWQ and MDIQ shall be specified in the executed FSS Agreement. FSS rendered by Seller under this Section shall consist of:

RCA No. 733	Fourth Revision	Sheet No.	15	RECEIVED
	Cancelling	Shoot No	45	NOV 30 2021
	Third Revision	Sheet No	15	

-			
	a)	The receipt of Gas on behalf of Customer at the Point(s) of Injection/Withdrawal at daily quantities up to the MDIQ plus Seller's Injection/Withdrawal Fuel Use;	
	b)	The storage of Gas in quantities not to exceed the MSQ, except as provided in Section 4.8;	
	c)	The tender of Gas by Seller for redelivery to or for the account of Customer at the Point of Injection/Withdrawal at daily quantities up to the MDWQ; and	
	d)	The receipt of Gas on behalf of Customer and redelivery of Gas for the account of Customer in excess of its applicable MDIQ and MDWQ on an interruptible basis.	
	e)	The MDIQ and MDWQ Ratchets set out in a Customer's FSS Agreement shall be based on the Customers Storage Quantity as a percentage of MSQ.	т
4.6	Selle the Cust unde Mon a po Capa	Rates and Charges: The amounts which shall be paid by Customer to er for each Month during the period of service hereunder shall include sum of the charges due under this Section that are applicable to tomer for such Month, computed by use of the applicable rates set forth er the Rate Schedule FSS (Section 35.1) which are effective during such th or portions thereof. If, at initiation of service, service is provided for only ortion of a Month, any applicable FSS Reservation Rate and FSS acity Rate shall be prorated for the number of Days that service is ided.	

January 14, 2022

RCA No. 733 Fourth Revision Cancelling Third Revision Sheet No.

16

16

Sheet No.

January 31, 2020

RECEIVED

STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

Cook Inlet Natural Gas Storage Alaska, LLC

,	/al	a) FSS Reservation Rate: As set forth on the Rate Schedule FSS, shall be paid each Month for each Mcf of Customer's Contract Withdrawal Quantity stated in its FSS Agreement, subject to adjustment for as provided in Section 34.2.	a)
	ity	b) FSS Capacity Rate: As set forth on the Rate Schedule FSS, shall be paid each Month for each Mcf of Customer's Total Contract Quantity stated in its FSS Agreement, subject to adjustment as provided in Section 34.2.	b)
	is of le of C	c) FSS Injection/Withdrawal Commodity Rate: As set forth on the Rate Schedule FSS, shall be paid each Month for each Mcf of Gas which is delivered to or for the account of Customer by Seller and each Mcf of Gas Customer delivers or causes to be delivered (less the applicable Injection/Withdrawal Fuel Use Percentage) at the Point(s) of Injection/Withdrawal during the Month. Such charges shall be applicable both on injection and on withdrawal.	c)
	of of T	d) Excess Storage Service Charge: As set forth on the Rate Schedule FSS, shall be applied to each Day that the Storage Quantity exceeds 100% of the Customer's Maximum Storage Quantity pursuant to Section 4.8 of the Rules and Regulations. The Excess Charge will be indicated on the Customer's monthly statement.	d)
	ge at as T	e) Seller's Injection/Withdrawal Fuel Use Charge: Customer shall furnish the Gas for Seller's Injection/Withdrawal Fuel Use in the provision of storage services as set out in Section 30.2 of this Tariff. Seller may agree that Customer may purchase from Seller equivalent Fuel Use quantities of Gas to cover Customer's obligation for the Seller's Injection/Withdrawal Fuel Use.	e)

				RECEIVED
RCA No. 733	Second Revision Cancelling	Sheet No.	17	NOV 03 2023
	First Revision	Sheet No.	17	STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

DECEIVED

Cook Inlet Natural Gas Storage Alaska, LLC

f) Overrun Storage Service Charge: As set forth in the Rate Schedule FSS, shall be paid for each Mcf of service provided on behalf of Customer pursuant to Section 4.7 of the Rules and Regulations. g) Rate Schedule FSS rates may be adjusted after operation of the CINGSA Facility to reflect that the physical performance of the CINGSA Facility is less than the Base Storage Design resulting in the design capacities set forth in the FSS Agreements being reduced. Such adjustment will be effective concurrent with such reductions in the FSS Agreements. Rate Schedule FSS rates shall be adjusted up or down from time to time h) outside of a general rate case filing to reflect changes in CINGSA's FSS capacity, but only to the extent that the change in capacity is incorporated into contracted FSS Agreements with existing Customers that have been approved by the RCA and/or into an FSS Agreement with a new Customer that has been approved by the RCA. Seller will provide Customers notice of proposed changes under this Section at least forty-five (45) days in advance of implementing the change. These adjustments shall be made on Customers' bills as a surcharge or a credit until incorporated into an RCA approved or accepted rate change. On Rate Schedule FSS, "Initial Capacity Tariff Rates" apply to Initial i) С Capacity Customers, and pursuant to Section 29.3c), "2023 Expansion С Tariff Rates" apply to the Customers with 2023 Expansion FSS С Agreements. С

E December 18, 2023

RCA No. 733 Second Revision Cancelling First Revision Sheet No.

Sheet No.

18

18

RECEIVED

January 31, 2020

STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

L

Cook Inlet Natural Gas Storage Alaska, LLC

- 4.7 Overrun Storage Service: Customer may request Seller to provide Overrun Storage Service under this Section 4.7 for quantities of Gas delivered in excess of Customer's MDIQ or MDWQ. Service requested under this Section Т must be nominated separately as "overrun" by Customer. Seller may provide Т such Overrun Storage Service on an interruptible basis if, in Seller's Т т reasonable judgment, Seller can provide the service on such Day without adverse effect on Seller's operations or on Seller's ability to meet higher Т priority obligations ("Authorized Overrun Storage Service"). Overrun Storage Т Т Service shall be scheduled by the Seller as provided in Section 18.3. Customer shall pay the Overrun Storage Service Charge pursuant to Т Section 4.6f) for such Overrun Storage Service and is subject to the Seller's Ν Ν Injection/Withdrawal Fuel Use Charge, pursuant to Section 4.6e).
- 4.8 Excess Storage Service: Customer may request Seller to provide Excess Ν Storage Service under this Section 4.8 for quantities of Gas in excess of Ν Customer's MSQ. Service requested under this Section must be nominated Ν separately as "excess" by Customer. Seller may provide such Excess Storage Ν Service on an interruptible basis if, in Seller's reasonable judgment, Seller can Ν provide the service on such Day without adverse effect on Seller's operations Ν or on Seller's ability to meet higher priority obligations. Excess Storage Ν Service shall be scheduled by the Seller as provided in Section 18.3. Ν Customer shall pay the Excess Storage Service Charge pursuant to Section 4.6d) for such Excess Storage Service. Т
- 4.9 Rules and Regulations: Any rates, terms, and conditions not specifically set forth in this Section 4, this Tariff or in Rate Schedule FSS may be provided in the FSS Agreement between the Customer and Seller and are subject at all times to approval by the Regulatory Commission of Alaska.

L – Section 4.6f) previously published on the First Revision of Sheet 18 has been moved to First Revision of Sheet 17 and renumbered as Section 4.6h).

Issued By: Cook Inlet Natural Gas Storage Alaska, LLC

RCA No. 733	Third Revision	Sheet No.	19	RECEIVED
100/1100 / 000	Cancelling	oncer no.	10	January 31, 2020
	Second Revision	Sheet No.	19	STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

SECTION 5	INTERRUPTIBLE STORAGE SERVICE (ISS) TERMS AND CONDITIONS
5.1	T Availability: Commencing on the Commencement Date, Interruptible Storage Service (ISS) is available to any Customer qualifying for natural gas storage service to the extent that:
	 Seller has determined that sufficient storage capacity is available to provide the service requested by the Customer; and
	 Seller and Customer have executed an agreement for storage service under this Section.
5.2	Applicability and Characteristics of Service: This Section 5 shall apply to all ISS which is rendered by Seller for Customer pursuant to an executed ISS Agreement under this Section. ISS rendered by Seller under this Section shall consist of the receipt or delivery of Gas for purposes of storing or loaning of Gas on behalf of Customer at the Point of Injection/Withdrawal up to the Maximum Storage Quantity (MSQ) at daily quantities up to the Maximum Daily Injection Quantity (MDIQ) plus Seller's Injection/Withdrawal Fuel Use and the tender of Gas by Seller for redelivery to or for the account of Customer at the Point of Injection/Withdrawal at daily quantities up to the Maximum Daily Withdrawal Quantity (MDWQ). The loaning of gas shall mean that the Customer creates a negative Storage Quantity balance and is required to replace such balance at some time in the future.
	a) Storage Service rendered under this Section shall be interruptible, and shall be available only when the Gas or capacity is not being used for injection, storage and withdrawal of higher priority services. Seller shall have the right to, and shall interrupt at any time, ISS under this Section as necessary to provide FSS under Section 4. Such ISS shall be offered in accordance with the provisions established in these Rules and Regulations.
	b) Upon notification from Seller, Customer may be required to cease or reduce deliveries to, or receipts from, ISS during the Day to protect FSS or to comply with the parameters of Customer's ISS Agreement. Further, Customer may be required to return Gas quantities and/or remove stored quantities upon notification from Seller. Such notification may be via e- mail or

DECENTED

RCA No.	First Revision	Sheet No. 20	RECEIVED
733 -	Cancelling Original	Sheet No. 20	NOV 3 0 2012
			STATE OF ALASKA Regulatory commission of Alaska

Cook Inlet Natural	Gas Storage Alaska, LLC	
	posting on Seller's website. Seller's notification shall specify the time frame within which stored quantities shall be removed and/or Gas quantities shall be returned, consistent with Seller's operating conditions, but in no event shall the specified time be sooner than the next Day after Seller's notification. In the event that the specified time for removal or return of Gas quantities is the next Day, the time of required removal or return shall begin from the time that Customer receives actual notice from Seller. Notices required after business hours for the next Day will be provided to Customer via e-mail or posting on Seller's website. In the event that Customer makes a timely nomination in response to notification by Seller to remove stored quantities and/or return Gas quantities, the obligation of the Customer to comply with the notification shall begin when Seller schedules the nomination.	
c)	In the event that the Customer is required to return Gas quantities and/or T remove stored quantities or such quantities are not delivered or received in accordance with the parameters outlined in any existing ISS. Agreement, Seller and Customer may mutually agree to an extended time frame and/or modified terms of such Agreement. In the event that Seller and Customer are unable to come to such agreement, Seller shall notify Customer and Customer shall deliver or receive the Gas quantities within the time frame specified in Seller's notice, which in no instance shall be less than one (1) Day. Any stored quantities not removed within the time frame specified by Seller's notice shall become the property of Seller at no cost to Seller, free and clear of any adverse claims. Any Gas quantities not returned within the time frame specified by Seller paid to obtain substitute gas.	
d)	In the event that stored quantities remain in Seller's system and/or Gas T quantities have not been returned to Seller's system at the termination of any existing ISS Agreement, Seller and Customer may mutually agree to an extended time frame and/or modify the terms of such agreement. In the event that Seller and Customer are unable to come to such agreement, Seller shall notify Customer	
	D	
D-Deleted two lin	nes inadvertly duplicated from Sheet 21	
TA8-733	Effective December 24, 2012	
-	Inlet Natural Gas Storage Alaska, LLC Starring Title: President	_

M. Colleen Starring

•

RCA No. 733Fourth Revision
Cancelling
Third RevisionSheet No.21RECEIVEDDanuary 31, 2020

STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

Cook Inlet Natural Gas Storage Alaska, LLC

and Customer shall remove the stored quantities and/or return the Gas quantities within the time frame specified by Seller's notice, which in no instance shall be less than one (1) Day. Any stored quantities not removed within the time frame specified by Seller's notice shall become the property of Seller at no cost to Seller, free and clear of any adverse claims. However, if Customer was unable to withdraw its ISS Storage Quantity due to an interruption of Customer's withdrawal service by Seller during the last 14 Days before termination, Seller will provide Customer a grace period following the expiration of the Agreement of one Day for each Day Customer was unable to withdraw Gas due to said interruption. In addition, any Gas quantities not returned within the time frame specified by Seller's notice shall be sold to Customer at 150% of the price Seller paid to obtain substitute gas.

- 5.3 ISS Rates and Charges: The amounts which shall be paid by Customer to Seller T for each Month during the period of service hereunder shall include the sum of the charges due under this Section that are applicable to Customer for such Month, computed by use of the applicable rates set forth under the Rate Schedule ISS (Section 35.2) which are effective during such Month or portions thereof.
 - a) ISS Rate: As set forth on Rate Schedule ISS, shall be paid each Month T on both the maximum quantity of Gas which is stored for the Customer for T the Month and the maximum quantity of Gas which is loaned to the T Customer during the Month, pursuant to Customer's ISS Agreement.
 - Т b) ISS Injection/Withdrawal Commodity Rate: As set forth on Rate Schedule ISS, shall be paid each Month for each Mcf of Gas which is delivered to or for the account of Customer by Seller and each Mcf of Gas Customer delivers delivered (less or causes to be the applicable Fuel Injection/Withdrawal Use Percentage) at the Point of Injection/Withdrawal during the Month. Such charges shall be applicable both on injection and on withdrawal.
 - c) Seller's Injection/Withdrawal Fuel Use Charge: Customer shall furnish the Gas for Seller's Injection/Withdrawal Fuel Use in the provision of storage services as set out in Section 30.2 of this Tariff. Seller may agree that Customer may purchase from Seller Equivalent Fuel Use quantities of Gas to cover Customer's obligation for the Seller's Injection/Withdrawal Fuel Use.

RCA No. 733	Second Revision	Sheet No.	22	RECEIVED
	Cancelling First Revision	Sheet No.	22	January 31, 2020
	r li st këvision	Sheet NO.	~~~	STATE OF ALASKA

- 5.4 Rate Schedule: ISS rates are subject to adjustment either upwards or downwards in the same manner as in Section 4.6g) and 4.6h) above.
- 5.5 Rules and Regulations: Any rates, terms, and conditions not specifically set forth in this Section 5, this Tariff or in Rate Schedule ISS may be provided in the ISS Agreement between the Customer and Seller and are subject at all times to approval by the Regulatory Commission of Alaska.

REGULATORY COMMISSION OF ALASKA

23

23

Sheet No.

NOV 30 2021 STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

RECEIVED

Cook Inlet Natural Gas Storage, Alaska, LLC

SECTION 6 TERM

6.1 The term of service under this Tariff shall be as set forth in the executed FSS or ISS Agreement.

SECTION 7 EXPIRATION OF TERM

- 7.1 Any FSS Customer's Gas remaining in the CINGSA Facility when that Customer's FSS Agreement expires shall be deemed sold to CINGSA at a price equal to 80% of the price realized for such Gas. However, if an FSS Customer was unable to withdraw its Gas quantities due to an interruption of Customer's withdrawal service by CINGSA during the last one hundred (100) Days before termination, Customer shall be allowed to withdraw its Gas within a period of time equal to the number of Days T withdrawal service was interrupted following the end of the FSS Agreement term.
 - a) Payment for the remaining quantities of Gas shall appear as a credit on the last statement rendered by CINGSA to Customer. To the extent that the credit exceeds the total charges in the statement, the difference shall be paid by CINGSA to Customer on or before the 25th Day of the Month following the Month of such expiration or termination.
 - b) In the event that CINGSA purchases and takes title to any of Customer's Gas pursuant to this Section, CINGSA shall dispose of such Gas using a process designed to maximize the proceeds recovered and shall credit 80% of the proceeds received against any amounts that may be owed by Customer to CINGSA.
- 7.2 Any ISS Customer's Gas remaining in the CINGSA Facility when that Customer's ISS Agreement expires shall be handled in accordance with Section 5.2d).

RCA No. 733Second Revision
Cancelling
First RevisionSheet No.2424

RECEIVED April 13, 2020

STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

Cook Inlet Natural Gas Storage, Alaska, LLC

SECTION 8	FACILITIES FOR ADDITIONAL SERVICES	Т
8.1	Unless otherwise agreed to by the Parties, Seller shall not be required to construct and install any facilities in addition to those required to perform the storage services that are the subject of existing FSS Agreements.	
8.2	"Facilities for Additional Services" means facilities requested by an FSS Customer to perform additional services such as to improve the reliability or service provided subject to an existing FSS Agreement that do not increase the design capacity of the CINGSA Facility. Requests to increase the design capacity of the CINGSA Facility are provided for in Section 29.	T T T T
	a) The Seller may enter into an agreement with a Customer, or a group of Customers, whereby the Seller will construct and install Facilities for Additional Services requested by Customer(s), and Seller may require that the Customer(s) reimburse Seller for all of Seller's costs of such construction and installation, either on a lump sum or incremental fee basis as agreed to by the Parties or as otherwise set forth in these Rules and Regulations. Prior to entering into such an agreement, the Seller will provide reasonable notice and invite all FSS Customers to participate.	T T T T T
	 b) Agreements for Facilities for Additional Services are subject to RCA approval as special contracts under RCA regulations. 	T T
ECTION 9	POINT(S) OF INJECTION/WITHDRAWAL	С
9.1	The Point(s) of Injection/Withdrawal, as defined in Section 2.57, are as follows:	С
	 a) KBPL Point of Injection/Withdrawal: The point of interconnection between Seller's CINGSA Facility and the KBPL (also known as the "415" meter). 	C C C
	 b) APC/ENSTAR Point of Injection/Withdrawal: The point of interconnection between Seller's CINGSA Facility and the APC/ENSTAR Line (also known as the "715" meter). 	C C C
9.2	Customer shall be solely responsible for making all arrangements and paying for the transportation of the Gas to be stored under the Agreement to the Point(s) of Injection/Withdrawal.	с
9.3	Seller shall have no obligation to receive deliveries from Transporter(s) for storage hereunder, or to redeliver quantities of Gas to Transporter(s) for the account of Customer hereunder, if Transporter(s) is (are) unwilling or unable to make or receive such deliveries or redeliveries, as the case may be.	C C C

RCA No. 733 Second Revision Cancelling First Revision Sheet No.

25

25

Sheet No.

January 31, 2020 STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

RECEIVED

Cook Inlet Natural Gas Storage Alaska, LLC

- SECTION 10 QUALITY OF GAS
 - 10.1 Heating Value of Gas:
 - a) The unit of quantity for the purpose of determining total heating value shall be one (1) cubic foot of anhydrous Gas at a temperature of sixty degrees Fahrenheit (60°F) and an absolute pressure of fourteen and sixty five onehundredths (14.65) Psia, dry.
 - b) Gas shall have a gross heating value of not less than nine hundred fifty (950) Btus per cubic foot and not more than one thousand fifty (1,050) Btus per cubic foot.
 - c) The gross heating value of Gas shall be determined from a representative composite Gas sample taken at the Point(s) of Injection/Withdrawal by periodic tests to be conducted monthly by the Seller or at such other intervals as the Parties may mutually agree. The determination shall be made by means of a Gas chromatograph using Gas Processors Association Publication 2145-09, as it may be revised, entitled "Table of Physical Properties for Hydrocarbons and Other Components of Interest to the Natural Gas Industry."

RCA No. 733 First Revisio	ion	Revision	First	733	No.	RCA
---------------------------	-----	----------	-------	-----	-----	-----

Sheet No. 26

Cancelling Original

Sheet No. 26

RECEIVED

FEB 1 6 2012

STATE OF ALASKA Regulatory commission of Alaska

Cook	Inlet	Natural	Gas	Storage	Alaska,	LLC
_						

- 10.2 Freedom from Objectionable Matter:
 - a) Gas shall be commercially free of dust, gum, gum-forming constituents, bacteria, or other liquid, gaseous, or solid matter which might interfere with its merchantability or cause injury to or interference with proper operation of the equipment through which it flows. Any substance that might become separated from the T Gas in transportation shall not exceed one hundred twenty T degrees Fahrenheit (120°F), shall not be odorized and shall not contain:

More than four (4) pounds of water vapor per million cubic feet of Gas;

More than one-quarter (0.25) grain of hydrogen sulfide per one hundred (100) cubic feet of Gas;

More than twenty (20) grains of sulfur per one hundred (100) cubic feet of Gas;

In excess of 1.0% by volume of oxygen; and/or

Т

In excess of 1.0% by volume of carbon dioxide.

10.3 Company's Right to Refuse Gas: Company shall have the right to accept or refuse to accept delivery of any Gas failing to meet the quality requirements of this Section 10.

TA 4-733

By:

Effective

APRIL 9, 2012

Issued By: Cook Inlet Natural Gas Storage Alaska, LLC,

Title: Vice President

M. Colleen Starring

RCA No. 733 First Revision

Sheet No. 27

Cancelling Original

Sheet No. 27



FEB 1 6 2012

STATE OF ALASKA Regulatory commission of Alaska

Cook Inlet Natural Gas Storage Alaska, LLC SECTION 11 MEASUREMENT AND MEASUREMENT EQUIPMENT 11.1 The volume of Gas delivered to Seller hereunder or redelivered to a) or for the account of Customer hereunder shall be measured at the Point of Measurement as follows: When the measuring equipment is an Ultrasonic meter, it shall be designed, installed, maintained and operated as recommended in the latest issue of American Gas Association ("AGA") Measurement Committee Transmission Report No. 9. "Measurement of Gas by Multipath Ultrasonic Meters," as such publication may be revised from time to time. When the measuring equipment is an orifice meter, the flow of Gas through the meter shall be computed in the manner recommended in AGA Report No. 3, "Orifice Metering of Natural Gas and Other T Related Hydrocarbon Fluids," properly using all factors set forth T therein. When the measuring equipment is a turbine meter, the volume of Gas delivered through the meter shall be computed in the manner recommended in AGA Report No. 7, "Measurement of Natural Gas by T Turbine Meter," properly using all factors set forth therein. Т When the measuring equipment is a positive displacement meter, the volume of Gas delivered through the meter shall be computed by properly applying correction factors for (1) absolute static Т T pressure, (2) flowing Gas temperature, and (3) compressibility ratio T to the volume delivered at flowing Gas pressures and T temperatures b) Auxiliary measuring equipment shall be installed, maintained and operated in accordance with generally accepted industry practices. APRIL 9, 2012 Effective TA 4-733

Issued By: Cook Inlet Natural Gas Storage Alaska, LLC,

M. Collin Itarring Title: Vice President

M. Colleen Starring

By:

RCA No. 733	First Revision	Sheet No. 28	RECEIVED
	Cancelling Original	Sheet No. 28	FEB 1 6 2012
Cook Inlet Nat	ural Gas Storage Alaska, LL	C	STATE OF ALASKA Regulatory commission of Alask
	The volume of Gas re- online Gas Flow Co volumetric measuremer	ceived or delivered shall be computer ("GFC"). Temperatunts shall be input into the GFC	re, pressure and T C at least once per

Mete				of "Flow 21 of the ma			-		
	· · ·			American P					
Cas	injected	or	withdrawn	hereunder	shall b	ne com	puted	usina	the

a) The unit of volume for the purpose of measurement shall be one thousand cubic feet of Gas at a temperature of sixty degrees Fahrenheit (60°F) and a pressure of fourteen and sixty five onehundredths(14.65) Psia, dry. The Dekatherm equivalent of such T unit of volume shall be determined by multiplying each such unit of volume by the total heating value per cubic foot of the Gas delivered hereunder (adjusted to a common temperature and pressure base) and by dividing the result by one thousand (1,000).

- b) The average absolute atmospheric (barometric) pressure at the Point of Injection/Withdrawal shall be assumed to be equal to fourteen and sixty five one-hundredths (14.65)Psia.
- c) The flowing temperature of the Gas shall be determined by means of an instrument of standard manufacture accepted in the industry for this purpose.

TA 4-733

By:

Effective

APRIL 9, 2012

T

Issued By: Cook Inlet Natural Gas Storage Alaska, LLC,

M. Collen Starring Title: Vice President

M. Colleen Starring

RCA No. 733	Second Revision	Sheet No.	29	RECEIVED
	Cancelling First Revision	Sheet No.	29	January 31, 2020
	FIRST REVISION	Sheet NO.	29	STATE OF ALASKA

- d) The density and compressibility shall be determined by a chromatographic analysis using the methods described in AGA Transmission Measurement Committee Report No. 8, "Compressibility Factors of Natural Gas and Other Related Hydrocarbon Gases," gross method, latest revision, at intervals of not more than six (6) Months, by means of an instrument of standard manufacture accepted in the industry for this purpose using a sample of Gas from the Gas stream at the Point(s) of Injection/Withdrawal.
- 11.3 All flow, measuring, testing and related equipment shall be of standard manufacture and type approved by the Seller. If applicable, Seller or Customer may install check measuring equipment, provided that such equipment shall be so installed as not to interfere with the operations of the Operator, as defined below. Seller, or Customer, in the presence of the other Party, shall have access to measuring equipment at all reasonable times, but the reading, calibrating, and adjusting thereof, if any, shall be done by the Operator. Seller or Customer shall have the right to be present at the time of the installing, reading, cleaning, changing, repairing, inspecting, testing, calibrating or adjusting performed by the Operator on the measuring equipment. The records from such measuring equipment shall remain the property of the Operator, but upon request, the Operator shall provide a requesting Party records, including charts, if any, together with calculations therefrom for inspection, subject to return within thirty (30) Days after receipt thereof. Reasonable care shall be exercised in the installation, maintenance and operation of the measuring equipment so as to avoid any inaccuracy in the determination of the volume of Gas injected and withdrawn. The accuracy of all measuring equipment shall be verified by Operator at reasonable intervals, and if requested, in the presence of representatives of a requesting Party, but neither Seller nor Customer shall be required to verify the accuracy of such equipment more frequently than once in any thirty (30) Day period. However, if an electronic measurement system is utilized, its components shall be calibrated at least once every ninety (90) Days.

REGULATORY COMMISSION OF ALASKA

Sheet No. 30

Cancelling Original RECEIVED

FEB 1 6 2012

STATE OF ALASKA REGULATORY CONMISSION OF ALASKA

Cook Inlet Natural Gas Storage Alaska, LLC

If any Party at any time desires a special test of any measuring equipment, it will promptly notify the Seller. The Seller shall then cooperate to secure a prompt verification of the accuracy of such equipment. Transportation and related expenses involved in the testing of meters under this Section 11.3 shall be borne by the Party incurring such T expenses.

The Operator, for purposes of this section, shall be the owner of the equipment referenced herein, or the agent of such owner, or such other person as the Parties may agree in writing.

If, upon any test, Operator's measuring equipment is found to be in error. such errors shall be taken into account in a practical manner in computing the deliveries. If the resultant aggregate error in the computed receipts or deliveries is not more than one percent (1%) for chromatograph or calorimeter and one percent (1%) for other measuring equipment, then previous receipts or deliveries shall be considered accurate. All equipment shall, in any case, be adjusted at the time of test to record correctly. If, however, the resultant aggregate error in computing receipts or deliveries exceeds one percent (1%) for chromatograph or calorimeter and one percent (1%) for other measuring equipment, then at a recording corresponding to the hourly average rate [m1] of Gas flow rate for the T period since the last preceding test, the previous recordings of such equipment shall be corrected to zero error for any period which is known definitely or agreed upon; if not definitely known, such correction shall be for a period extending over one-half of the time elapsed since the date of the last test.

11.4 In the event any measuring equipment is out of service, or is found to be T registering inaccurately and the error is not determinable by test, previous recordings of injections or withdrawals through such equipment shall be determined as follows; provided, however, that the correction period shall not exceed one (1) year:

TA 4-733

By:

Effective

APRIL 9, 2012

Issued By: Cook Inlet Natural Gas Storage Alaska, LLC,

M. Lollen Starring Title: Vice President

M. Colleen Starring

RCA	No.	733	First Revision
ł			

Cancelling

Original

Sheet No. 31

31 Sheet No.

RECEIVED

FEB 1 6 2012

STATE OF ALASKA							
REGULATORY	COMMISSION	OF ALASKA					

Cook Inlet Natural Gas Storage Alaska, LLC By using the registration of any check meter or meters if installed a) and accurately registering; or b) In the absence of 11.4a), by correcting the error if the percentage of error is ascertainable by calibration, special test or mathematical calculation; or In the absence of both 11.4a) and 11.4b), then by estimating the C) quantity of receipt or delivery based on receipts or deliveries during preceding periods under similar conditions when the meter was registering accurately. If at any time during the term hereof, a new method or technique is 11.5 developed with respect to Gas measurement or the determination of the factors used in such Gas measurement, such new method or technique may be substituted upon mutual agreement thereto by the Parties. 11.6 The Parties shall preserve all test data, charts, if any, and other similar T records for a period of at least three (3) years or such longer period as T may be required by public authority. Т In accordance with the provisions of Sections 11.3 and 11.4 above, Seller 11.7 will use the best information available to close its allocation of quantities for a service Month. For the purposes of this Section 11.7, "close" shall mean five (5) Business Days after the applicable service Month. To the extent that adjustments are made after the date of such close, such adjustments ("Prior Period Adjustments" or "PPA") shall be treated in accordance with this Section 11.7. If the PPA are due to the correction of measurement data or reallocation of volumes, such adjustments shall be processed within two (2) years of the applicable service Month. Such deadline shall not apply in the case of deliberate omission or misrepresentation or mutual mistake of fact. The Parties' other statutory or contractual rights shall not be diminished by this standard. APRIL 9, 2012 Effective TA 4-733

Issued By: Cook Inlet Natural Gas Storage Alaska, LLC,

By: M. Loller Starring Title: Vice President M. Colleen Starring

					RECEIVED
RCA No.	733	Original Cancelling	Sheet No. Sheet No.	32	JAN 1 9 2011
			Sheet No.		STATE OF ALASKA Regulatory commission of Alasi
Cook Inlet N	latural (Gas Storage Alaska, LLC	;		
SECTION ·	12 FC	RCE MAJEURE AND	REMEDIES		
12.1	deliv Part not I of its by s requ circu resu ordir Maje Tarif reas Part oblig satis	ept as otherwise provid very of a notice claim y asserting Force Maje be liable for any delay s obligations under this such Force Majeure E ired to cure or rem umstance or event, o lting from the failure hary care of a reasona eure Event with respect of, "Force Majeure E onable control of a P- y unable, wholly or in pation, covenant, or ofying the requirement eure Events" shall in- ints:	ing a valid Force M eure, a Party providir in, or failure of, its p a Tariff, if such delay Event until such time edy such Force Ma r combination or co of a Party to exercise ble and prudent ope of to such Party. For Event" means any arty that directly or in part, to perform of condition in this T s and definition set f	lajeure Even og such noti performance or failure is e as is real ajeure Even ontinuation se the stan rator is not the purpose event beyon ndirectly re- or comply wo forth above,	nt by a ce shall e of any caused sonably nt. Any thereof, idard of a Force e of this ond the inders a vith any oject to "Force
	a)	eruption, storm, hu	htning, landslide, ea rricane, hurricane w losion, or well blowo	arning, floo	
	b)	public enemy, v insurrection, riot, e	other industrial distu war, military ope epidemic, arrest or ority, terrorist act, ci ;	ration, blo restraint	ockade, by any
	c)	or lines of pipe, and replacement, test, equipment, facilities	dent to machinery, e the repair, maintena or alteration to , or lines of pipe, an or the partial or enti	the mag the freezi	vement, chinery, ing of a
Pursuant to	U-10-05	51(9)	Effective: Janu	uary 31, 2011	

Issued By: Cook Inlet Natural Gas Storage Alaska, LLC By: <u>Alaste determined</u> Title: Vice President

M. Lollen Starring

CINGSA Proposed Tariff filed January 19, 2011 U-10-51(9) Appendix D Page 33 of 77

M. Colleen Starring

RCA No.	733	Original Cancelling	Sheet No. Sheet No.	33	RECEIVED JAN 1 9 2011 STATE OF ALASKA REGULATORY GOMMISSION OF ALASKA
Cook Inlet N	latural G	Gas Storage Alaska, LLC	>		
	d)	Act, order, or requ	iisition of any gover al authority, or any		
12.2	of a shal	ess otherwise provided ny Force Majeure Ev I not, relieve the other I ding any payment obl	ent asserted by a P Party of its obligatio	arty does n	ot, and
12.3	Notio	ce and Remedy.			
The	Party c	laiming Force Majeur	e under Section 12 w	vill:	
	a)	after its occurrence	rty of the Force Maje e, giving reasonably the time required to	full particula	ars and
	b)	Keep the other developments; and	Party informed	of all sig	nificant
	c)	Majeure Event an Agreement as soor the settlement of s	in good faith to d resume full perfo n as reasonably prac trikes, lockouts, or o f a failed Gas well w affected Party).	rmance und ticable (exce other labor d	der this ept that lisputes
12.4	and that more	ring the period from the thereafter, a Party cla the Force Majeure Ev than six (6) months to agree on a com	iming a Force Majer rent will not be reme , the Parties will me	ure Event es died for a pe et within thi	timates eriod of rty (30)

during the period of the Force Majeure Event that is consistent

Pursuant to U-10-051(9)

Effective: January 31, 2011

Issued By: Cook Inlet Natural Gas Storage Alaska, LLC

with the intent of this Tariff.

By: M. Loller Starring

1

M. Colleen Starring

Title: Vice President CINGSA Proposed Tariff filed January 19, 2011 U-10-51(9) Appendix D Page 34 of 77 RCA No. 733Third RevisionSheet No.34Cancelling
Second RevisionSheet No.34

FEB 22 2021

RECEIVED

STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

Cook Inlet Natural Gas Storage, Alaska, LLC

SECTION 13 DELIVERY PRESSURE

- 13.1 For injections, Customer shall contract with Transporter(s) to require Transporter(s) to deliver Gas to Seller at Transporter(s)'s available pipeline pressure at the Point(s) of Injection/Withdrawal, but in no event at a pressure less than five hundred fifty (550) psig nor greater than eight hundred fifty (850) psig unless otherwise mutually agreed to by Seller, Customer and Transporter. Seller will not accept volumes for injection at the APC/ENSTAR Line Point of Injection/Withdrawal without its prior written approval.
- 13.2 For withdrawals via the KBPL Point of Injection/Withdrawal, Seller shall redeliver Gas to Transporter for the account of Customer at a pressure sufficient to enter Transporters pipeline system at the KBPL Point of Injection/Withdrawal, but in no event at a pressure in excess of nine hundred seventy five (975) psig or less than six hundred (600) psig unless otherwise mutually agreed to by Seller, Customer and Transporter.
- 13.3 For withdrawals via the APC/ENSTAR Point of Injection/Withdrawal, Seller shall redeliver Gas to Transporter for the account of Customer at a pressure sufficient to enter Transporter's pipeline system at the APC/ENSTAR Point of T Injection/Withdrawal, but in no event at a pressure in excess of one thousand fifty (1,050) psig or less than six hundred (600) psig unless otherwise mutually agreed to by Seller, Customer and Transporter.

RCA No. 733	Second	Sheet No.	35	RECEIVED
	Cancelling First	Sheet No.	35	NOV 1 4 2014
	•	-		STATE OF ALASKA REGULATORY COMMISSION OF ALASKA
Cook Inlet Nat	ural Gas Storage	Alaska, LLC		
an fa bhaird Richan ba fa la la bhairte ann ann a sha ann an ann ann ann ann ann ann ann an	BILLING AND I		anga kananga kanga ka	
14.1	shall render (for (a) postmarked electronic data applicable) an in preceding Month shall be based of at points where quantities. Seller injection from o Month, in place reasonably avail	fore the ninth (9th) Business r purposes of this Section or (b) time-stamped and ele mechanism to the desig nvoice to Customer setting for n under the applicable Rate S on actuals (if available) or bes e OBAs exist shall be invo may utilize estimates of the r redelivered to or for acco e of actual quantities when lable, provided that adjustmet rences between such estimate	14.1, "rend ectronically nated site, orth the amo Schedule(s). st available biced based quantity of C bunt of Cus actual qua ents shall be	er" shall mean transmitted via whichever is bunt due for the Seller's invoice data. Quantities N on scheduled N Gas received for N tomer during a antities are not e made in later
		on necessary for invoicing pu omer shall furnish such inforn ay of the Month.		
	books, records a the accuracy of	Customer have the right to ex and charts of the other to the f any invoice, charge or co of the provisions hereof.	e extent neo	cessary to verify
14.2	Day after the da Due Date"). If th Day, payment v date. Payments transfer directe otherwise agree number specifie payment differs	mer shall pay any invoice on ate the invoice was delivered be invoice payment due date vill be due on the first Busin by Customer to Seller shall d to a bank account des ed to by the Parties. Custom d by Seller to which the paym from invoiced amount, remitt at, except when the payment i	I to the cust does not fa ness Day fo be made in signated by ner shall ide nent relates. ance detail s	omer ("Payment II on a Business ollowing the due the form of wire Seller, unless entify the invoice In the event the shall be provided
TA 18-733		Effective	Decembe	r 29, 2014

Issued By: Cook Inlet Natural Gas Storage Alaska, LLC By: /s/ Moira K. Smith Title: Vice Pres

Title: Vice President and General Counsel

Moira K. Smith

, 1, ,

RCA No. 733 First Revision

Sheet No. 36

Cancelling Original

36 Sheet No.



FEB 1 6 2012

STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

Cook Inlet Natural Gas Storage Alaska, LLC

funds transfer (EFT), in which case, the remittance detail is due within two Business Days of the payment due date.

Should Customer fail to pay all of the amount of any invoice as herein provided when such amount is due, interest on the unpaid portion of the invoice shall accrue from the due date until the date of payment at an annual rate of interest equal to the prime rate plus two percent (2%) charged by the Federal Reserve Bank during that period, but which in no event shall be higher than the maximum rate permitted by applicable law. If such failure to pay continues, then following thirty (30) Days prior written notice from Seller of its intent to abandon service under the Agreement, Customer shall be deemed to have consented to such abandonment of service, unless within the thirty (30) Day period Customer pays to Seller the entire balance due with interest, except for any amounts which Customer may dispute in good faith.

If Customer fails to pay any amount when due in accordance with Section 14.2 other than amounts disputed in good faith by Customer, Seller, in addition to any other remedy it may have hereunder, may suspend further injection or withdrawal of Gas for Customer and may enter into Agreements to provide service to others using Customer's capacity and deliverability.

In the event Customer in good faith disputes the amount of any such invoice or part thereof, it shall pay to Seller such amounts not in dispute and shall furnished to Seller documentation supporting the basis for the dispute within thirty (30) Days of a demand for payment made by Seller. T Within thirty (30) Days of final resolution of a payment dispute, Customer shall remit to Seller payment of any amounts determined to be owed as a result of such final resolution along with any interest accrued thereon calculated in accordance with this Section 14.2.

The foregoing shall be in addition to any other remedies Seller may have, including confiscation of all or a portion of Customer's Storage Quantity, at law or in equity, with respect to Customer's failure to pay the amount of any invoice not otherwise disputed in good faith.

TA 4-733

Effective A

APRIL 9, 2012

Issued By: Cook Inlet Natural Gas Storage Alaska, LLC,

By: M. Colleen Starring M. Colleen Starring Title: Vice President

RECEIVED

RCA No.	733	Original	Sheet No. 37	JAN 1 9 2011
		Cancelling		
			Sheet No.	STATE OF ALASKA
				CULATORY COMMISSION OF ALASKA

Cook Inlet Natural Gas Storage Alaska, LLC

14.3 Adjustment of Invoicing Errors: Subject to the provisions of Section 11 above of these Rules and Regulations, if it shall be found that at any time or times Customer has been overcharged or undercharged and Customer shall have actually paid the invoice containing such charges, then within thirty (30) Days after the final determination thereof, either Seller shall refund the amount of any such overcharge or Customer shall pay the amount of any such undercharge. In the event an error is discovered in the amount invoiced in any invoice rendered by Seller, such error shall be adjusted within thirty (30) Days of the determination thereof, provided that claim therefore shall have been made within thirty (30) Days from the date of discovery of such error, but in any event within six (6) Months from the date of such invoice. The Party receiving such request for adjustment shall have three (3) Months to rebut such claim, otherwise the invoice shall be adjusted as requested. The preceding time limits do not apply to deliberate omission or misrepresentation or mutual mistake of fact or government-required rate changes. The Parties' statutory or contractual rights shall not otherwise be diminished by this Section. If the Parties are unable to agree on the adjustment of any claimed error, any resort by either of the Parties to legal proceedings shall be commenced within fifteen (15) Months after the supposed cause of action is alleged to have arisen, or shall thereafter be forever barred.

Pursuant to U-10-051(9)

Effective: January 31, 2011

Issued By: Cook Inlet Natural Gas Storage Alaska, LLC

By: M. Lollen Starring

M. Colleen Starring

Title: Vice President CINGSA Proposed Tariff filed January 19, 2011 U-10-51(9) Appendix D Page 38 of 77

	700				RECEIVED
RCA No.	733	Original	Sheet No	. 38	JAN 1 9 2011
		Cancelling	Sheet No		JAN 1 0 2011
			Glicerivo	•	STATE OF ALASKA REGULATORY COMMINISION OF ALASI
Cook Inlet I	Vatural	Gas Storage Alaska, LL	С		
SECTION	15 D	EFAULT, SUSPENSIO	ON AND TERMINAT	ION	
15.	of I peti inso sub Eve cov Agr mai gua of f Sec Agr	ent of Default: All of th Default") under this Ta ition for bankruptcy olvency, reorganizatio oject of an involuntary ent"); (b) if a Party fa renant or obligation reement; or (c) if a fa terial covenant or o aranty provided pursua this Tariff. All referen- ction 15 shall mean reement, as the case eements or terms and	ariff: (a) if a Party or , winding-up, liqu n or other similar p bankruptcy petition ils to perform or br imposed upon it Party's Guarantor f bligation imposed ant to the creditwor ces to "Service Ag the applicable FSS e may be, and an	tits Guarante idation, dis proceeding, (each, a "Ba eaches any t under a fails to perfe upon it un thiness requ preement" un Agreement y other door	or files a solution, or is the inkruptcy material Service orm any der any irements nder this t or ISS cuments,
15.2	res con opti	mination of Service: I pect to a Party ("De itinuing, the other Pa ion, terminate any Se ty by proceeding as fo	efaulting Party") ha irty ("Non-Defaulting ervice Agreement v	as occurred J Party") ma	and is ay, at its
				(in second s	
Pursuant to	o U-10-0	051(9)	Effective: Ja	nuary 31, 2011	1

Issued By: Cook Inlet Natural Gas Storage Alaska, LLC By:

M. Coller Starring

ľ.

 $\left(\cdot, \cdot \right)^{2}$

M. Colleen Starring

Title: Vice President CINGSA Proposed Tariff filed January 19, 2011 U-10-51(9) Appendix D Page 39 of 77

RCA No.	733	Original	Sheet No. 39	RECEIVE
		Cancelling		JAN 1 9 2011
			Sheet No.	
				STATE OF ALASKA Regulatory commission of A

DECEN

Cook Inlet Natural Gas Storage Alaska, LLC

- The Non-Defaulting Party shall provide written notice (by a) email, promptly followed by courier or overnight mail) to the Defaulting Party, stating specifically the cause for terminating the Service Agreement(s) and declaring it to be the intention of the Non-Defaulting Party to terminate the same on a designated date not earlier than thirty (30) Days thereafter (the "Early Termination Date"). Except when a the Event of Default is a Bankruptcy Event, the Defaulting Party shall have thirty (30) Days after receipt of such notice to remedy or remove the cause or causes stated in the notice. If the Event of Default is a Bankruptcy Event, the termination shall be effective when the notice is delivered. If within the thirty (30) Day period, the Defaulting Party removes and remedies said cause or causes and fully indemnifies the Non-Defaulting Party for any and all consequences of such Event of Default, subject to the provisions of Section 24, by good and sufficient means acceptable to the Non-Defaulting Party, on or before the Early Termination Date, then such notice shall be withdrawn and the Service Agreement shall continue in full force and effect:
- b) If the Defaulting Party does not remedy and remove the cause or causes and does not indemnify the Non-Defaulting Party for any and all consequences of such Event of Default within the thirty (30) Day period (other than in the case where the Event of Default is a Bankruptcy Event), then, after any necessary notice to regulatory bodies having jurisdiction (which notice may be given at the same time as the termination notice to Defaulting Party), the Service Agreement(s) shall be terminated as of the Early Termination Date, provided that notice of termination has not been withdrawn prior thereto;

Pursuant to U-10-051(9)

Effective: January 31, 2011

Title: Vice President

Issued By: Cook Inlet Natural Gas Storage Alaska, LLC

By: M. Lollen Starring

M. Colleen Starring

CINGSA Proposed Tariff filed January 19, 2011 U-10-51(9) Appendix D Page 40 of 77

				RECEIVED
RCA No.	733	Original	Sheet No. 40	
		Cancelling		JAN 1 9 2011
			Sheet No.	STATE OF ALASKA Regulatory commission of Alaska

Cook Inlet Natural Gas Storage Alaska, LLC

- c) Any termination of such Service Agreement(s) pursuant to the provisions of this Section 15.2c) shall be without prejudice to the right of Seller to collect any amounts then due to it for services rendered prior to the time of termination, and shall be without prejudice to the right of Customer to receive any service for which it has paid and is entitled but has not received prior to the time of termination; and
- d) Any termination of such Service Agreement(s) shall be without prejudice to or waiver of any other right or remedies, at law or in equity, to which the Non-Defaulting Party may be otherwise entitled: including, without limitation, the right to seek damages (subject to Section 24 below) and/or off-set any amount owing by the Defaulting Party to the Non-Defaulting Party under the Service Agreement(s) against any amount owing by the Non-Defaulting Party to the Defaulting Party under the Service Agreement(s) or under any other agreement.
- 15.3 Without limiting the Non-Defaulting Party's right to terminate the Service Agreement under Section 15.2 and any other remedies that may be available to the Non-Defaulting Party as the result of an Event of Default, after expiration of any notice period contemplated in Section 15.2a) and a failure during any such period to remove or remedy the Event of Default, a Non-Defaulting Party shall have the right to suspend performance of the Service Agreement with respect to the Defaulting Party during any such Event of Default.

Pursuant to U-10-051(9)

Effective: January 31, 2011

Issued By: Cook Inlet Natural Gas Storage Alaska, LLC

M. Lollen Starring

By:

M. Colleen Starring

Title: Vice President CINGSA Proposed Tariff filed January 19, 2011 U-10-51(9) Appendix D Page 41 of 77

RCA No.	733	Original	Sheet No. 41	RECEIVED
And the second of the second		Cancelling		JAN 1 9 2011
			Sheet No.	STATE OF ALASKA REGULATORY COMMINISION OF ALA
Cook Inlet	Natural	Gas Storage Alaska, LLC		

SECTION 16 NON-WAIVER OF FUTURE DEFAULTS

No waiver by either Party of any one or more defaults by the other in the performance of any provisions of the Agreement shall operate or be construed as a waiver of any future default or defaults, whether of a like or of a different character.

Pursuant to U-10-051(9)

January 31, 2011 Effective:

Issued By: Cook Inlet Natural Gas Storage Alaska, LLC By:

M. Edlen Starring

M. Colleen Starring

Title: Vice President CINGSA Proposed Tariff filed January 19, 2011 U-10-51(9) Appendix D Page 42 of 77

RCA No. 733	S
	C
	F

Second Revision Cancelling First Revision Sheet No.

Sheet No.

42

42

OCT 2 6 2017

RECEIVED

STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

Cook Inlet Natural Gas Storage Alaska, LLC

SECTION 17 REQUESTS FOR STORAGE SERVICE

- 17.1 Requests for Generally Available (Unsubscribed) Storage Service:
 - a) Requests: To request FSS or ISS service, a potential Customer shall submit a request for such service consistent with the applicable Tariff requirements. Seller shall evaluate and respond to such requests within twelve (12) Business Days, and shall begin service, if an Agreement is executed, as soon as is reasonably possible. Such a request shall be considered acceptable only if the information specified in this Section 17.1 is provided in writing, but Seller may waive all or a portion of such information in individual instances. Requests for service shall be sent to:

Cook Inlet Natural Gas Storage Alaska, LLC 3000 Spenard Road, Anchorage Alaska 99503

b) Form of Requests: Customers requesting FSS and ISS service must provide the following information:

Customer(s) names:

Name, address, representative, telephone number, fax number and email address of Customer(s).

Type of service(s) requested:

Service desired (i.e., FSS or ISS).

Quantity (stated in Mcfs):

Maximum Storage Quantity (which, unless otherwise agreed to, shall not be less than one hundred thousand (100,000) Mcf).

Contract Injection Quantity.

Contract Withdrawal Quantity.

С

RCA No. 733

First Revision Cancelling Original Sheet No.

Sheet No.

January 31, 2020

RECEIVED

STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

Cook Inlet Natural Gas Storage Alaska, LLC

Term of service:

Date service is requested to commence: Unless agreed to otherwise, service shall commence on April 1 of each Contract Year.

43

43

FSS shall terminate on March 31 of the last Contract Year.

c) Subsequent Information:

Customer must comply with all of Seller's Creditworthiness requirements as set forth in Section 26 below of the Rules and Regulations herein. After receipt of a Request for service hereunder, Seller may require that Customer furnish additional information necessary for Seller to maintain operational integrity of its gas storage facility as a prerequisite to Seller offering to execute an Agreement with Customer. Such information may include proof of Customer's ability to transport from the Point(s) of Injection/Withdrawal Gas that it has delivered to Seller and that Seller has redelivered to the Point(s) of Injection/Withdrawal. Customer may redact competitively sensitive information from documents furnished hereunder.

d) Request Validity: Customer's Request for storage service shall be considered null and void if Seller has tendered an Agreement for execution to Customer, and Customer fails to execute that Agreement within fifteen (15) Days. In determining whether it is feasible to tender an Agreement relative to FSS and ISS, Seller will assess available capacity, after provision for existing services, operating constraints and pending requests for service. Seller will not execute an Agreement which relates to requests for such service for which it does not have sufficient available capacity.

RCA No. 733	Second Revision	Sheet No.	44	RECEIVED
	Cancelling			January 31, 2020
	First Revision	Sheet No.	44	STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

Cook Inlet Natural Gas Storage Alaska, LLC

SECTION 18 NOMINATIONS, SCHEDULING AND ALLOCATIONS

- 18.1 Nominations:
 - a) Seller will accept nominations for Storage Service as provided herein. A valid nomination is a data set which contains any Seller-required data elements such as Customer name, contract number, type of service, the flow rate and the daily quantities of Gas that Customer desires to be injected or withdrawn from storage by Point(s) of Injection/Withdrawal.
 C Seller will not accept nominations for injections at the APC/ENSTAR Line
 C Point of Injection/Withdrawal without its prior written approval. Seller will
 C accept nominations electronically. The daily nomination shall be in a format and form approved by the Seller. If the Seller has developed an electronic form and format for the submission of daily nominations, it may require its use by Customers.

Each nomination shall indicate whether it is being submitted as a Standard or as an Intra-Day nomination. The standard flow rate for nominations, confirmations and scheduling shall be Mcf/d.

Seller, at its sole option, may require direct confirmation of the Customer's supplier and interconnecting pipeline arrangements.

Customer may use an agent to provide all or a portion of its nomination data, provided that Seller is so advised in advance in writing. A Customer that uses an agent for such nomination purposes shall hold Seller harmless for all actions or inactions of its agent.

(The next Sheet is Sheet 44.1)

RCA No. 733	Original	Sheet No 44.1	RECEIVED
	Cancelling	Sheet No.	MAR 3 0 2012
		Sheet Ito.	

STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

Cook Inlet Natural Gas Storage Alaska, LLC

Standard Nominations: A "Standard Nomination" is a nomination b) for storage service for any Day. The Standard Nomination shall include a begin date and end date, which must be within the term of the Customer's Service Agreement. Each Day within such date range nomination shall be considered an original nomination. Subsequent nominations for one (1) or more Days within the range shall supersede only the Days specified. The Days outside the range of the subsequent nomination are unaffected. Nominations have a prospective effect only. Seller shall process all new or revised nominations that are received by 4:00 p.m. ACT on the C Day before the applicable Service Day. Customer may nominate zero (0) for a daily quantity. However, Seller may decline to provide service for any hour if the total net nominations by all C Customers for that hour do not exceed 5,000 Mcf/d. Customer C shall also inform Seller in advance of each Month of the desired order of priority of injections and withdrawals under each Agreement and Seller may rely thereon (or in the absence of such information, upon Seller's judgment) if allocation under such Agreement is required.

TA 4-733

By:

Effective

APRIL 9, 2012

Issued By: Cook Inlet Natural Gas Storage Alaska, LLC,

M. Colleen Starring

Title: Vice President

RCA No. **Second Revision** Cancelling

45 Sheet No.

45 Sheet No.

RECEIVED OCT 2 3 2012

STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

Cook Inlet Natural Gas Storage Alaska, LLC

First Revision

Standard Nominations Timetable: c)

> The timetable for a Standard Nominations shall be as follows on the Day before a Service Day:

> > 4:00 p.m. ACT for receipt of nominations by Seller;

12:00 a.m. ACT (next Day) for flow of Gas.

The nomination must include the flow rate (expressed in increments of 100 Mcf/d) for each hour of the Day and total volume (expressed in Mcf and rounded to the nearest 100 Mcf) for the Day.

The Injection/Withdrawal Fuel Use Percentage described in T Section 30 of this tariff will be taken out of the Customers' nominations.

After the end of each Day, Seller shall make available the final scheduled quantities for the just completed Day.

Seller, as receiver of nominations, initiates the confirmation process. The Party receiving a request for confirmation or an unsolicited confirmation response may waive the obligation of the Seller to initiate the confirmation process. The sending Party will adhere to nomination, confirmation and scheduling deadlines. The Party receiving the communication shall have the right to waive any deadline, on a non-discriminatory basis.

(The next Sheet is Sheet 45.1)

TA 8-733

Effective December 24, 2012

Issued By: Cook Inlet Natural Gas Storage Alaska, LLC By: M. Collin Starring Title: President

M. Colleen Starring

733

RCA No. 733	Original		Sheet No.	45.1	DEACH	69 00 (10)
	Cancelling		Sheet No. 45.1	RECEIV	/ED	
		_	Sheet No.	43.1	MAR 3 0	2012
L					STATE OF ALA Regulatory commissi	SKA On of Alaska
Cook Inlet Natur	ral Gas Storage A	laska, LLC				
	d) Departu	res from Nomination	ns			N
	Custom	nominations remain er shall use reasor r anticipated signific	nable efforts	s to notify the	e Seller of any	
	nominat confirme change be char	ller has the right to tion, that is not in ed by the relevant s data. The Seller al nged or the Seller m Gas flows.	n an appro supplier, an lso has the	oved format, d/or that wou right to requir	has not been ld retroactively e a nomination	N
		(The next Sheet is S	Sheet 46)			
			Efforting			
TA 4-733			Effective	APRIL 9, 2012		

Issued By: Cook Inlet Natural Gas Storage Alaska, LLC,

By: M. Lollan Starring Title: Vice President M. Colleen Starring

NOV 30 2021

46

46

Cancelling **First Revision**

Sheet No.

STATE OF ALASKA **REGULATORY COMMISSION OF ALASKA**

RFCFIVFD

Cook Inlet Natural Gas Storage, Alaska, LLC

- 18.2 Intra-Day Nominations:
 - Any nomination by eligible Customers submitted after the standard a) nomination deadline shall be an Intra-Day Nomination. An Intra-Day Nomination shall be effective for one (1) Day only. The nomination process set forth in Section 18.1a) above shall also apply to the Т Intra-Day Nominations. An Intra-Day nomination shall be a revised daily rate.

Customer shall provide a minimum of three (3) hours' notice prior to the effective time for Gas to flow.

An Intra-Day nomination shall be subject to Transporter's confirmations and Seller's operating conditions. If Transporter's confirmation is not received, the Intra-Day Nomination will not be accepted. Seller will not accept a reduced Intra-Day Nomination for any quantity deemed already delivered based on an average hourly flow or for a time period that has already passed.

- Seller will accept Intra-Day Nominations for Storage Service on a b) best efforts basis.
- For purposes of providing notice of any nomination changes to a c) Customer and or Customer's agent, Seller shall contact either party by telephone or other instant communication device. With respect to changes initiated by Seller, if a Customer so elects, Customer may provide a telephone number and Seller will contact Customer at that number to alert Customer that a change has been made; provided that where an interruptible Customer nomination is superseded by a firm Customer's Intra-Day Nomination, Seller shall provide notice of such supersession to the interruptible Customer in the same manner that Seller uses to notify Customers of OFOs.

TA46-733	Effective:	January 14, 2022
Issued By: Cook Inlet Natural Gas Storage	, Alaska, LLC	
Ву:	Title:	

RCA No. 733	Third Revision	Sheet No.	47	RECEIVED
	Cancelling Second Revision	Sheet No.	47	FEB 22 2021
				- STATE OF ALASKA

STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

Cook Inlet Natural Gas Storage, Alaska, LLC

- 18.3 Scheduling of Storage and Allocation of Service: For each Day, Seller will schedule injections and withdrawals of Gas, on the basis of: Storage nominations made by Customers (which Seller is hereby authorized to rely upon in its scheduling); Storage capacity available on Seller's system in light of nominations and requests: and overall operating conditions from time to time. If, on any Day, Seller determines that the capacity of its system is insufficient to serve all Storage nominations scheduled for such Day, or to accept the quantities of Gas tendered, capacity shall be allocated to provide service in the following order:
 - a) In scheduling FSS nominations on any Day when capacity is constrained, Seller shall allocate service on a pro rata basis based upon each Customer's contracted CIQ or CWQ volumes compared to the total contracted daily injection or withdrawal volumes of all Customers nominating volumes on such Day.
 - b) In scheduling nominated quantities for FSS Overrun Storage Service or ISS hereunder; after providing for FSS service, and to the extent there is remaining capacity, Seller shall allocate service ratably. A Customer's share of remaining capacity is calculated as a Customers' good faith nominations for FSS Overrun Storage Service and ISS divided by the total good faith nominations for FSS Overrun Storage Service and ISS.
 - In scheduling nominated quantities for FSS Excess Storage Service or c) ISS hereunder; after providing for FSS service, and to the extent there is remaining capacity, Seller shall allocate service ratably. A Customer's share of remaining capacity is calculated as a Customers' good faith nominations for FSS Excess Storage Service and ISS divided by the total good faith nominations for FSS Excess Storage Service and ISS.

D D D D D D

D D

D – Deleted duplicate Sections 18.4 and 18.5.

April 8, 2021

RCA No. 733	Second Revision	Sheet No.	48	RECEIVED
	Cancelling			January 31, 2020
	First Revision	Sheet No.	48	STATE OF ALASKA

Cook Inlet Natural Gas Storage Alaska, LLC

- 18.4 Delivery of Gas: Seller, subject to the other provisions hereof, shall make daily L delivery, to the extent practicable, of Equivalent Quantities of Gas at the Point(s) L,C of Injection/Withdrawal.
- 18.5 Hourly Variation: Injections and Withdrawals shall be made at uniform rates to L the extent practicable.
- 18.6 Limitation on Obligation: Should the quantities of gas received from customer(s) by seller at the Point(s) of Injection/Withdrawal exceed the Maximum Daily C Injection Quantity plus the Seller's Injection/Withdrawal Fuel Use, Seller shall notify Customer(s) of such fact, within a reasonable time after such becomes known, and Customer(s) shall seek to reduce deliveries to Seller forthwith. In the event any such excess delivery would jeopardize the safety of Seller's operations and/or its ability to meet its contract commitments to others, such decisions being solely within the judgment and discretion of Seller, Seller shall have the right to refuse to accept, without any-liability to Customer, or any other person, all or such part of said excess delivery as Seller deems necessary, and shall notify Customer accordingly.

L – Section 18.4 and 18.5 moved from First Revision of Sheet 47

REGULATORY COMMISSION OF ALASKA

	Second Revision Cancelling	Sheet No.	49	RECEIVED
RCA No. 733		Sheet No.		April 13, 2020
	First Revision	Sheet No.	49	STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

Cook Inlet Natural Gas Storage, Alaska, LLC

SECTION 19	OPERATIONAL INFORMATION
19.1	Capacity Information, Operational Bulletins, and Advisories:
	a) Each Day, Seller will provide information concerning the storage service capacity available from Seller, as well as operational bulletins and advisories to Customers with effective FSS or ISS Agreements.

SECTION 20 RIGHT TO COMMINGLE

Seller shall have the unqualified right to commingle Gas stored hereunder with Gas from all Customers and from other sources and treat and handle all such Gas without distinction as to source. It is recognized that Gas redelivered will not be the same molecules as those received.

С

Ν

N N N

					RECEIVED
RCA No.	733	_Original Cancelling	Sheet No.	50	JAN 1 9 2011
		Gancening	Sheet No.		STATE OF ALASKA
				- H	REGULATORY COMMISSION OF ALASK
Cook Inlet Na	itural G	as Storage Alaska, LLC			
SECTION 2	1 TIT	LE TRANSFER OF G	AS IN STORAGE		
21.1	recei	stomer (Transferor) m ved for storage by Se Customer (Transfered	ller for the account		
	a)	Both Transferor an provide Seller with v			
	b)	The purchase does its MSQ as specified			
	c)	In Seller's judgmen adverse effect on Se higher priority obliga	eller's operations or i		
21.2	comp With (24)	r will recognize tra outing available Storag drawal Quantities on hours after receiving on 21.1a) above.	e Quantity and appl a prospective basis	icable Inject within twe	ion and nty-four
Pursuant to L	J-10-05	1(9)	Effective: Jan	uary 31, 2011	

Issued By: Cook Inlet Natural Gas Storage Alaska, LLC By: M.A. Vice President

M. Collen Starring

M. Colleen Starring

CINGSA Proposed Tariff filed January 19, 2011 U-10-51(9) Appendix D Page 51 of 77 RCA No. 733

First Revision Cancelling Original Sheet No.

51 51

Sheet No.

January 31, 2020

RECEIVED

STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

Cook Inlet Natural Gas Storage Alaska, LLC

SECTION 22 RESPONSIBILITY FOR GAS

Upon receiving delivery of Gas for storage at the Point(s) of Injection/Withdrawal, Seller shall **C** be in exclusive control and possession of such Gas and responsible for any loss thereof, or any and all injury or damage caused thereby. When the Gas has been redelivered at the Point(s) of Injection/Withdrawal Customer shall be in exclusive control and possession of such **C** Gas and responsible for any and all injury or damage caused thereby.

SECTION 23 WARRANTY OF TITLE

Customer shall warrant for itself, its successors, and assigns, that it has, or will have, at the time of delivery of the Gas for injection hereunder good title to such Gas and/or good right to cause the Gas to be delivered to Seller for storage. Customer shall warrant for itself, its successors, and assigns, that the Gas it warrants hereunder shall be free and clear of all liens, encumbrances or claims; that it will indemnify and save Seller harmless from all suits, actions, debts, accounts, damages, costs, losses, and expenses arising from or out of adverse claims of any and all persons to said Gas, and/or to royalties, taxes, license fees, or charges thereon which are directly applicable to such delivery of Gas and that it will indemnify and save Seller harmless from all taxes or assessments which may be directly levied and assessed upon such delivery and which are by law payable and the obligation of the party making such delivery.

700	Original	Obset No. 50	112021-2-
733	Original	Sheet No. 52	JAN 1 9 2011
	Cancelling		JAN I J ZON
		Sheet No.	STATE OF ALASKA
			STATE OF ALASKA REGULATORY GOMMISSION OF ALASKA
	733		Cancelling

RECEIVE

Cook Inlet Natural Gas Storage Alaska, LLC

SECTION 24 LIMITATION ON LIABILITY

Notwithstanding any other provision in the Agreement or this Tariff, neither Party shall be liable to the other Party under the Agreement, the Rules and Regulations or Rate Schedules for any special, exemplary, indirect, incidental, punitive or consequential damages (including, without limitation, lost profits, loss of revenues, or loss of business), of any nature, however arising, whether arising in tort, contract, indemnity, strict liability or otherwise and regardless of whether such Party is or has been made aware of the possibility of such damages (including lost profits).

SECTION 25 RESPONSIBILITY FOR ASSOCIATED TRANSPORTATION

The transportation of quantities to be stored hereunder to and from the Point of Injection/Withdrawal is solely the Customer's responsibility. The inability of third-party pipelines to receive or deliver Gas to or from the CINGSA Facility shall not be deemed to constitute non-availability of service or establish any basis of liability for Seller. Customer is not relieved from its obligations under its Agreement as of the Commencement Date for failure to secure transportation to and from the **CINGSA** Facility.

Pursuant to U-10-051(9)

Effective:

Title: Vice President

January 31, 2011

Issued By: Cook Inlet Natural Gas Storage Alaska, LLC

By: A. Lollen Starreng

1 4

M. Colleen Starring

CINGSA Proposed Tariff filed January 19, 2011 U-10-51(9) Appendix D Page 53 of 77

RCA No. 733	Original	Sheet No. 53	RECEIVED
100/110. 100	Cancelling		JAN 1 9 2011
	Canceling	Sheet No.	JAN 1 3 2011
			STATE OF ALASKA
			REGULATORY COMMISSION OF ALASKA
Cook Inlet Natural	Gas Storage Alaska, LLC		
SECTION 26 CF	REDITWORTHINESS		
failu brea cred acco alter belo prov Sect will I	tomer shall establish ar ire by Customer to main ach of the Agreement. ditworthiness in the contex ordance with the criteria s rnative, or if Customer is n ow) by Seller, creditwor viding and maintaining a tion 26.3 below. Customer be based upon the level of Guarantor will be considere	tain creditworthiness re Seller shall evaluate at of all services purcha et forth in this Section ot deemed Creditworthy thiness may be esta Credit Alternative, as , or its Guarantor's, cred	epresents a Customer's ased and in 26.1. As an (as defined ablished by defined in litworthiness
a)	at least "BBB-" by Sta McGraw Hill Companie by Moody's Investors least "BBB-" by Fitch Shippers not rated by S a rating of at least "BI Service ("Rating Agent (low), the Seller shall which must be stab Creditworthy. If Custom	unsecured debt securitie indard & Poor's, a divi s, Inc. "S&P" and/or at I Service, Inc. "Moody's Ratings (Fitch), or fo itandard & Poor's, Fitch, BB(low)" by Dominion E cies"). If the rating is B also consider the out ole or better to be her is rated by more that he lowest rating will be	sion of The least "Baa3" ", and/or at or Canadian or Moody's, Bond Rating BB- or BBB clook rating, considered n one of the
b)	senior unsecured debts is at least "BBB-" by S8 rating is BBB- or BBB(the outlook rating, whi considered Creditworthy by more than one of the rating will be taken into does not have one of 26.1b), then Customer	or does not have rated a securities, its long-term i &P or "Baa3" by Moody' low), the Seller shall all ch must be stable or b y. If Customer or Guara le Rating Agencies, only account. If Customer of f the ratings in Section or Guarantor has obta from a recognized an	ssuer rating s, and if the so consider petter to be ntor is rated y the lowest or Guarantor n 26.1a) or ained, at its
Pursuant to U-10-05	51(9)	Effective: January 31, 2	2011

Issued By: Cook Inlet Natural Gas Storage Alaska, LLC By: M.A. Vice President

M. Collen Starring

1.1

CINGSA Proposed Tariff filed January 19, 2011 U-10-51(9) Appendix D Page 54 of 77

M. Colleen Starring

No.	733	Original
		Cancelling

RCA

Sheet No. 54

JAN 1 9 2011

RECEIVED

Sheet No.

STATE OF ALASKA Regulatory commission of Alaska

Cook Inlet Natural Gas Storage Alaska, LLC

debt rating agency acceptable to Seller that is at least equivalent to the minimum ratings set forth in Section 26.1a) or 26.1b); and

- c) Customer's estimated Total Remaining Obligations under its FSS or ISS Agreement are less than Customer's and or its Guarantor's Tangible Net Worth. For the purposes of this Tariff, the term "Total Remaining Obligations" shall mean the then NPV of the remaining payment obligations, exclusive of commodity rates under all of a Customer's FSS Agreements with CINGSA plus an amount necessary to collateralize all of a Customer's ISS Agreements. For the purposes of this Tariff, "Tangible Net Worth" shall mean total assets, less total liabilities, less intangible assets, less off-balance sheet obligations. Intangible assets include, but are not limited to, goodwill, patents, and unamortized loan costs.
- 26.2 Creditworthiness Assessment:

If Customer or its Guarantor is not Creditworthy pursuant to Section 26.1a), 26.1b), or 26.1c), Customer may request that Seller evaluate Customer's or its Guarantor's creditworthiness based upon the level of service requested relative to Customer's current and future ability to meet its obligations. After its assessment of Customer's or its Guarantor's creditworthiness, Seller, in its reasonable discretion, may deem Customer or its Guarantor Creditworthy. Such creditworthiness assessment shall be based upon Seller's evaluation, in its reasonable discretion, of any or all of the following requested information and credit criteria:

 Audited financial reports: Customer's or its Guarantor's balance sheets, income statements, cash flow statements, notes to financial statements and auditor's opinions along with key ratios and trends regarding liquidity, tangible net worth, asset management, debt management, capital structure, operational efficiency, profitability trends, debt coverage, and overall financial condition;

Pursuant to U-10-051(9)

Effective: January 31, 2011

Issued By: Cook Inlet Natural Gas Storage Alaska, LLC

By: M. I Men Starring

Title: Vice President CINGSA Proposed Tariff filed January 19, 2011 U-10-51(9) Appendix D Page 55 of 77

M. Colleen Starring

RCA No.	733	Original	Sheet No.	55	RECEIVED
1.		Cancelling	Sheet No.		JAN 1 9 2011
				R	STATE OF ALASKA GULATORY GOMMISSION OF ALASKA
Cook Inlet N	atural G	as Storage Alaska, LLC			
	b)	The nature of Custor business of general conditions specific to	economic condition		and the second
	c)	Information from cre references;	dit rating agencies	and/or ban	k/trade
	d)	Prompt payment of basis and any other i in its analysis of Cust	nformation that Se	ller deems r	A STATE AND A STAT
	e)	Customer or its Gua federal, state or o judgment subject arrangement, adjustn receiver, trustee, o Guarantor's property;	ther applicable b to liquidation nent, composition, o r assignee of th	ankruptcy , reorgan or appointme	law or ization, ent of a
	f)	If Customer fails to Section 26.1 or Secti of receipt of such r Guarantor may be Seller.	on 26.2 within five equest from Selle	(5) Busines r, Customer	s Days or its
26.3	main Custo provi	eller determines that tained or been deeme omer has the option of ding to Seller one or r herein, a "Credit Altern	ed Creditworthy as receiving service more of the followi	described under this T	above, ariff by
	a)	An acceptable guarar Total Remaining Obli Gas related to Gas Rate Schedule from a a third party deemed with the criteria set fo	gations and, if app loaned or redelive a corporate affiliate Creditworthy by S	licable the v red under to of the Custo eller in acco	alue of he ISS omer or
Pursuant to	U-10-051	1(9)	Effective: Janu	uary 31, 2011	

Issued By: Cook Inlet Natural Gas Storage Alaska, LLC Title: Vice President

By: M. Collen Starring

5

1

M. Colleen Starring

CINGSA Proposed Tariff filed January 19, 2011 U-10-51(9) Appendix D Page 56 of 77

RCA No.	733	Original	Sheet No.	56	101 1 0 2014
		Cancelling			JAN 1 9 2011
			Sheet No.		STATE OF ALASKA
					REGULATORY COMMISSION OF ALASKA

RECEIVE

Cook Inlet Natural Gas Storage Alaska, LLC

b) An acceptable irrevocable standby letter of credit in an amount equal to the Customer's pro rata share of the undepreciated storage facilities as determined when the Customer becomes non-Creditworthy, subject to annual modifications, and up to three (3) months of FSS reservation and capacity charges, such letter of credit to be issued by a major commercial bank operating in the United States or Canada with an unenhanced senior unsecured debt rating of at least "A" by S&P and "A2" by Moody's, or up to three (3) months of estimated service charges and the value of Gas related to Gas loaned or redelivered under the ISS Rate Schedule. A Customer's pro rata share of the undepreciated storage facilities shall be the amount of Seller's outstanding net plant attributable to the Customer's ratable portion of the total MSQ of Seller calculated using the following equation: Customer pro rata share of undepreciated storage facilities = (Customer MSQ + Total Seller's MSQ) x Seller's outstanding net plant. As an example for illustrative purposes only, if Customer A, who has MSQ of 5.5 Bcf of the total 11 Bcf of Seller's FSS capacity (50%) becomes non-creditworthy under the provisions hereof in year 5 of its agreement, the Customer would be required to post, under this provision, an irrevocable standby letter of credit in an amount equal to 50% of the net plant of the Seller as of the most recent quarter ended financial statement. The net plant of Seller shall include all facility costs, including any base gas, less any accumulated depreciation. Thus, if the net plant of Seller equals \$125 million at the time Customer A becomes non-creditworthy, Customer A would be required to post a letter of credit in the amount of \$62.5 million. If Customer A remains non-creditworthy during the following year, and the net plant is reduced in that year by \$6 million in depreciation, Customer A's letter of credit can be reduced by 50% of the \$6 million net plant reduction, or \$3 million. The amount shall be recalculated and modified each year based on Seller's net plant balance; or

Pursuant to U-10-051(9)

Effective:

Title: Vice President

January 31, 2011

Issued By: Cook Inlet Natural Gas Storage Alaska, LLC

By: M. Collen Itaring

M. Colleen Starring

CINGSA Proposed Tariff filed January 19, 2011 U-10-51(9) Appendix D Page 57 of 77

				RECEIVED)
RCA No.	733	Original	Sheet No.	57 JAN 1 9 2011	
		Cancelling		JAN I U.LON	
			Sheet No.	STATE OF ALASKA REGULATORY COMMISSION OF A	ASKA
					L'IUIII
Cook Inlet Na	atural G	as Storage Alaska, LLC			
	c)	Any other credit alternative Customer and Seller. Such accepted on a non-discrime subsequent requests to such on a not unduly discrime Customer with a written ere request to substitute credit a	other credit a inatory basis bstitute such natory basis explanation	alternatives shall be is. Seller may deny h credit alternatives s and will provide of any denial of a	
26.4	shall Custo throu	e case of a new service agre be provided by Customer upo omer shall maintain its Creo gh provision of a Credit Alte Agreement and the ISS Agree	on execution ditworthiness ernative, duri	of the Agreement. s, either directly or	
	a)	With respect to the obligati Customer or Guarantor, a regulation by the Securitie Customer or Guarantor sha five (5) Business Days of adverse change in its bu (financial or otherwise) or it time during the term of th Guarantor notifies Seller, or own investigation, that th adverse change in the bu (financial or otherwise), or Customer or its Guarantor the, bank or financial institu	ions under the s applicable and Exchall notify Sell of the detail usiness, pro- results of op- ne Agreeme if Seller det here has b- usiness, pro- results of op- ceases to h	e, is not subject to hange Commission, ller in writing, within ils of any material operties, conditions perations. If, at any ent(s), Customer or termines through its been any material operties, conditions operations such that be Creditworthy, or	

Pursuant to U-10-051(9)

Effective: Ja

as a Credit Alternative ceases to maintain an unenhanced senior unsecured debt rating of at least "A" by S&P and "A2" by Moody's, or the Creditworthiness of the Customer or Credit Alternative is insufficient to fulfill its portion of the Customer's Credit Alternative requirement, Seller may request and Customer shall deliver to Seller within five (5) Business Days of such request a Credit Alternative in the

amount set forth in the Seller's request.

January 31, 2011

Issued By: Cook Inlet Natural Gas Storage Alaska, LLC By: July and Title: Vice President

By: M. Collen Starring

er e Litte

M. Colleen Starring

CINGSA Proposed Tariff filed January 19, 2011 U-10-51(9) Appendix D Page 58 of 77

RECEIVED

RCA No.	733	Original	Sheet No.	58	JAN 1 9 2011
		Cancelling			
			Sheet No.		STATE OF ALASKA
					REGULATORY COMMISSION OF ALASKA

Cook Inlet Natural Gas Storage Alaska, LLC

The Parties agree that the failure of Customer to remain b) Creditworthy or supply or maintain any Credit Alternative is an Event of Default under Section 15.1 and shall not; (i) relieve Customer of its other obligations under the Agreement(s); or (ii) prejudice Seller's right to seek performance under the Agreement(s). Upon notification by Seller that either Customer, or Guarantor no longer meets Seller's creditworthiness standards, or the bank or financial institution offering a Letter of Credit as a Credit Alternative no longer maintains an unenhanced senior unsecured debt rating of at least "A" by S&P or "A2" by Moody's, Customer shall within five (5) Business Days, pay for one (1) Month of service in advance to continue service, provided service has commenced. Customer shall in any event, within thirty (30) Days, provide an acceptable Credit Alternative consistent with Seller's creditworthiness standards. If the Customer fails to provide an acceptable Credit Alternative to Seller within these time periods, Seller may suspend its performance of obligations immediately. Any such suspension shall not relieve Customer from any obligation to pay any further charges or amounts payable to Seller under the Agreement(s). If Customer fails to provide an acceptable Credit Alternative after such suspension, Seller may, at its option at any time, in addition to any other remedy that may be available to it under the Agreement(s), at law, or in equity, upon ten (10) Days written notice to Customer and the RCA, terminate the Agreement(s). As part of resolution of Customer's obligations, Seller may net quantities of Gas in Seller's FSS and ISS accounts.

Pursuant to U-10-051(9)

Effective:

Title: Vice President

January 31, 2011

Issued By: Cook Inlet Natural Gas Storage Alaska, LLC

By: M. Edlen Starring

đ. .

M. Colleen Starring

CINGSA Proposed Tariff filed January 19, 2011 U-10-51(9) Appendix D Page 59 of 77

RCA No. 733	First Revision	Sheet No.	59	RECEIVED
	Cancelling			NOV 30 2021
	Original	Sheet No.	59	STATE OF ALASKA
				REGULATORY COMMISSION OF ALASKA

Cook Inlet Natural Gas Storage, Alaska, LLC

- c) If a Credit Alternative is provided pursuant to this Section 26.4 and Customer or its Guarantor, as applicable, is later determined by Seller to be Creditworthy, Seller shall return to Customer whatever Credit Alternative Seller then holds within five (5) Business Days of Customer's request. Upon expiration of the Customer's Agreement(s), unless such Credit Alternative has been returned pursuant to the preceding sentence, any remaining Credit Alternative will be returned after resolving any and all disputed invoice amounts under the expired Agreement(s).
- 26.5 Customer FSS Agreements are contracts under which Seller may extend financial accommodations to Customers, if a petition is filed, by or against Customer, any of its affiliates, or any Guarantor, under any chapter of the bankruptcy code of the United States or under legislation of a similar nature of any other nation. If Seller is not permitted by a bankruptcy court to reject the Agreement(s) as a result of such filing, Seller may consider the bankruptcy filing in determining whether Customer remains Creditworthy, and in determining what, if any, additional Credit Alternative must be submitted by or for Customer as a condition to Customer's creditworthiness under the Agreement(s).
- 26.6 The credit terms and conditions set forth above shall survive, subject to any required regulatory approvals, and shall continue in effect for the term(s) of the Agreement(s).

				RECEIVED
RCA No.	733	Original	Sheet No. 60	1411 1 0 004
		Cancelling	(*****	JAN 1 9 2011
			Sheet No.	STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

Cook Inlet Natural Gas Storage Alaska, LLC

SECTION 27 REGULATIONS

The Agreement(s) and the respective obligations of the Parties are subject to all valid laws, orders, rules and regulations of duly constituted authorities having jurisdiction, and are conditioned upon the issuance by any governmental agency having jurisdiction, of requisite authorizations for Seller to provide the storage service contemplated hereby, and for Transporter to make deliveries and redeliveries necessary to effect the storage service provided for in the Agreement(s). The Parties shall promptly file for and diligently pursue all governmental authorizations necessary for the implementation and maintenance of the Agreement(s).

Pursuant to U-10-051(9)

Effective: Jan

Title: Vice President

January 31, 2011

Issued By: Cook Inlet Natural Gas Storage Alaska, LLC

- By: M. Collen Starring
 - M. Colleen Starring

CINGSA Proposed Tariff filed January 19, 2011 U-10-51(9) Appendix D Page 61 of 77

RCA No. 7	33	Original	Sheet N	o. 61	RECEIVED
	55	Cancelling	Sheet N		JAN 1 9 2011
				10	STATE OF ALASKA Regulatory commission of Alask
Cook Inlet Na	tural Ga	as Storage Alaska, LL	С		
SECTION 28	R TRA	ANSFER AND ASSI	SNMENT		
28.1	Any consc Custo entitle	company which sh blidation to the pro omer or of Seller, a ed to the rights and	nall succeed by perties, substantia is the case may b shall be subject to	lly as an e be, shall if the obliga	entirety, of eligible be tions of its
	Custo of its obliga	cessor in title und omer. Either Seller of obligations under the ations under the Ag ted at the time of suc	or Customer may, le Agreement, ass reement to a com	without relie	eving itself ts rights or
28.2	have perma	thstanding any othe the right to assign anently release (in v 's consent if the:	(in whole or in pa	art) the Agr	eement or
	a)	assignee is a par to flow through to or payable by Cust		and all am	ounts paid
	b)	assignee meets requirements of Se			worthiness
	c)	Customer has pro advance of any ass			notice in
Pursuant to U	-10-051	(9)	Effective:	January 31, 20	111

M. Colleen Starring

1823

CINGSA Proposed Tariff filed January 19, 2011 U-10-51(9) Appendix D Page 62 of 77

R	E	CI	EI	V	E	D

			1.4.1.1		
JAN	-1	2	3	^	
1/1 1/1		6.8	.,	п	11
AIN	- 8	27			
01114	-		-	v	

Sheet No.

Sheet No.

62

STATE OF ALASKA Egulatory commission of Alaska

Cook Inlet Natural Gas Storage Alaska, LLC

Original

Cancelling

RCA No.

733

28.3 Seller may, as security for its indebtedness, assign, mortgage or pledge any of its rights or obligations under the Agreement, including its rights to receive payments, to any other entity, and Customer will execute any consent agreement with such entity and provide such certificates and other documents as Seller may reasonably request in connection with any such assignment, provided the terms of any such consent, certificates or other documents do not materially alter the rights and obligations of Customer under the Agreement and/or this Tariff. Customer also may assign or pledge the Agreement under the provisions of any mortgage, deed of trust, indenture or similar instrument which it has executed or may hereafter execute covering substantially all of its properties. Otherwise, except as set forth in this Section 28, neither Party shall assign the Agreement or any of its rights thereunder unless it first shall have obtained the consent thereto in writing of the other Party, which consent shall not be unreasonably withheld, conditioned or delayed.

Pursuant to U-10-051(9)

Effective: Ja

Title: Vice President

January 31, 2011

Issued By: Cook Inlet Natural Gas Storage Alaska, LLC

- By: M. Lollen Staring
 - M. Colleen Starring

CINGSA Proposed Tariff filed January 19, 2011 U-10-51(9) Appendix D Page 63 of 77

RCA No. 733	Second Revision	Sheet No.	63	RECEIVED
	Cancelling	oneer no.	00	January 31, 2020
	First Revision	Sheet No.	63	STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

Cook Inlet Natural Gas Storage Alaska, LLC

SECTION 29 CINGSA FACILITY EXPANSION AND EXPANSION CAPACITY RIGHTS

- 29.1 Process:
 - a) At any time following the completion of the initial capacity and provided that any prior expansion is completed, any Customer may request Seller to increase the design capacity of the CINGSA Facility up to the maximum physical capacity of the storage reservoir allowed by the Alaska Oil & Gas Conservation Commission ("Requested Expansion Capacity") by delivering a written notice of such request to Seller ("Notice of Request") not less than thirty (30) months prior to the date which the Customer desires any facilities necessary for all or part of such Requested Expansion Capacity to be in-service.
 - Upon receipt of a Notice of Request from a Customer for Requested b) Expansion Capacity, or if Seller determines that expansion is possible, С Seller will, on a commercially reasonable basis, solicit interest in С expansion capacity through public non-discriminatory open season С solicitations that may be either binding or non-binding in nature and may include minimally acceptable rate levels, subscription levels, and service terms that are consistent with existing approved Tariff provisions. Seller will also solicit interest from existing Customers that may wish to turn back any of their firm storage capacity. Seller will provide at least 30 Days' prior public notice of any such open season solicitation via public media designed to reach interested parties. Seller has the right to require bidders in such an expansion open season to comply with capacity request requirements set forth in Section 17.1 above of the Rules and Regulations. All binding open season bids meeting the requirements as stated in the open season notice and/or guidelines shall be evaluated on a non-discriminatory basis.

т

RCA No. 733 First Revision	Sheet No	64	- RECEIVED
Cancelling Original	Sheet No.	64	MAR 3 0 2012
			STATE OF ALASKA REGULATORY CORMISSION OF ALASKA

- Seller will then provide all Customer(s) and other expansion c) capacity open season participants ("Prospective Expansion Customer(s)"): (i) the amount of the Requested Expansion Capacity that Seller is willing and able, on a non-discriminatory basis, to offer to each Prospective Expansion Customer, including turnback capacity under Section 28.2 accepted by Seller for incorporation into the expansion project and the physical and technical limitations of the storage reservoir ("Allocated Expansion Capacity"); and (ii) the good faith anticipated facility design, capital T costs, and rate which it deems necessary to design, construct and Т operate the Allocated Expansion. The Allocated Expansion Capacity together with any expansion capacity requested by other parties in the open season shall be the "Expansion Project".
- Within thirty (30) Days of receipt of the Expansion Project design. d) cost and rate information from Seller, each Prospective Expansion Customer shall notify Seller in writing whether or not it is prepared to commit to and enter into both a binding Expansion Precedent T Agreement and a separate FSS Agreement ("Expansion FSS T Agreement") on mutually agreeable terms and conditions and for a mutually agreeable term with Seller. If a Prospective Expansion T Customer notifies Seller that it is prepared to enter into the Expansion Precedent Agreement and the Expansion FSS Agreement, Seller will in turn, after consultations and discussions with other Prospective Expansion Customer(s), notify Prospective Expansion Customer whether or not Seller is willing to proceed with the development of the Allocated Expansion Capacity. Seller shall proceed with the expansion if it is technically feasible and commercially reasonable. If Seller decides not to proceed with the requested expansion. Seller shall notify all Prospective Expansion Customers of the basis for that decision.

TA 4-733

Effective _

APRIL 9, 2012

Issued By: Cook Inlet Natural Gas Storage Alaska, LLC,

By: M. Loller Starring M. Colleen Starring Title: Vice President

RCA No. 733 First Revision	Sheet No. 65	RECEIVED
Cancelling Original	Sheet No. 65	FEB 1 6 2012
		STATE OF ALASKA Regulatory commission of Alaska
Cook Inlet Natural Gas Storage Alaska, LLC	C	

- e) If both Seller and Prospective Expansion Customer are prepared to enter into an Expansion Precedent Agreement and Expansion FSS Agreement and to proceed with the development of the Allocated Expansion Capacity on mutually agreeable terms, then Seller will prepare the Expansion Precedent Agreement and Expansion FSS Agreement which will incorporate the Expansion Project design, cost and rate information provided previously to Prospective Expansion Customer, modified to reflect then-current information and estimates.
- f) Prospective Expansion Customer shall execute the mutually agreeable Expansion Precedent Agreement delivered to it by Seller within thirty (30) Days of receipt of same and is thereafter T subject to the terms of the Expansion Precedent Agreement and T the Expansion FSS Agreement. If the Expansion Precedent Agreement and Expansion FSS Agreement are not executed by Prospective Expansion Customer within the periods specified, Seller will be under no obligation to continue with the development of the Allocated Expansion Capacity, or to enter into the Expansion Precedent Agreement or Expansion FSS Agreement.
- 29.2 Regulatory Authorizations:
 - a) Upon execution by Prospective Expansion Customer of an Expansion Precedent Agreement, Seller shall strive, if necessary, to file for approval of the Expansion Project capacity and rates with all applicable regulatory and government bodies not later than twenty-four (24) months prior to the in-service date of the Allocated Expansion Capacity identified in the Expansion Precedent Agreement and Expansion FSS Agreement.

TA 4-733

By:

Effective

APRIL 9, 2012

Issued By: Cook Inlet Natural Gas Storage Alaska, LLC,

M. Collen Itarring Title: Vice President

M. Colleen Starring

				- NEVEIVED
RCA No.	733	Original	Sheet No. 66	2
		Cancelling		JAN 1 9 2011
			Sheet No.	STATE OF ALASKA Regulatory commission of Alaska

Cook Inlet Natural Gas Storage Alaska, LLC

- b) The terms and conditions of the Expansion FSS Agreement shall be subject to, among other things, the outcome of the open season held as a result of Seller's receipt of the Notice of Request.
- c) Costs allocated to Expansion Project services will be rolled in with all CINGSA costs if the rates established thereby for existing CINGSA services would be lower than existing rates. Otherwise, costs will be allocated to Expansion Project services on an incremental basis. In such case, only the costs of the Expansion Project will be allocated to services associated with the Expansion Project.
- 29.3 Creditworthiness. Expansion Project customers will be subject to the creditworthiness provisions set out in the above-referenced Expansion Precedent Agreement and Expansion FSS Agreement in relation to the Allocated Expansion Capacity.

Pursuant to U-10-051(9)

Effective:

Title: Vice President

January 31, 2011

Issued By: Cook Inlet Natural Gas Storage Alaska, LLC

- By: M. Collen Starring
 - M. Colleen Starring

CINGSA Proposed Tariff filed January 19, 2011 U-10-51(9) Appendix D Page 67 of 77 RCA No. 733 Fourth Revision Cancelling Third Revision Sheet No.

Sheet No.

67

67

January 31, 2020 STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

Cook Inlet Natural Gas Storage Alaska, LLC

SECTION 30 FUEL USE PERCENTAGE

- 30.1 Fuel Use: The term "Fuel Use" shall mean the quantity of Gas required by Seller for injections into and withdrawals from the CINGSA Facility and shall be inclusive of (a) compressor fuel use Gas and (b) except for extraordinary losses for which Seller may separately seek RCA approval for recovery, lost-andunaccounted-for ("LAUF") Gas including, but not limited to, unrecoverable Gas and all losses of stored Gas resulting from operations of the field, reservoir or equipment failure, natural or forced migration of the storage Gas; provided however, Seller is barred from seeking recovery of LAUF in its rates or in any future revenue requirement filing if the proximate cause of the LAUF is an act or omission of Seller (including its employees, contractors, agents, or affiliates) which would constitute negligence. This provision does not bar CINGSA from seeking to apportion liability in any manner provided by law.
- 30.2 The term "Seller's Injection/Withdrawal Fuel Use" shall mean the quantity of Gas required by Seller for Fuel Use and shall be equal to the Injection/Withdrawal Fuel Use Percentage multiplied by the quantities tendered to Seller by Customer for injection at the Point(s) of Injection/Withdrawal.
- 30.3 Current Injection/Withdrawal Fuel Use Percentage: The Current Injection/Withdrawal Fuel Use Percentage shall be redetermined annually on or before May 15th to reflect the estimated Fuel Use and estimated injections for the ensuing twelve-month period beginning the first Day in July.



С

RCA No. 733 Fourth Revision Cancelling Third Revision Sheet No.

68 68

Sheet No.

January 31, 2020 STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

RECEIVED

Cook Inlet Natural Gas Storage Alaska, LLC

30.4	Curr	ent Fuel Use Surcharge Percentage:	
	a)	Seller shall annually compare the quantity of Fuel Use Gas retained by the Seller for the most recent Contract Year with the quantity of actual compressor Fuel Use incurred at the CINGSA Facility for the same period to determine any over/under collection of Fuel Use by the Seller.	
	b)	The Current Fuel Use Surcharge Percentage shall be determined by dividing this over/under collection quantity by the projected injections, as described in Section 30.3 above. Under-recovered Fuel Use will result in a positive adjustment and over-recovered Fuel Use will result in a negative adjustment.	
	c)	The Current Fuel Use Surcharge Percentage will be effective on the first Day of July.	
30.5	Perc Perc Perc Perc	ction/Withdrawal Fuel Use Percentage: The Injection/Withdrawal Fuel Use centage shall be the sum of the Current Injection/Withdrawal Fuel Use centage calculated in Section 30.3 and the Current Fuel Use Surcharge centage calculated in Section 30.4. The Injection/Withdrawal Fuel Use centage is the applicable percentage of Fuel Use, as in Section 35.3a), in shall be assessed on all FSS and ISS injections of Gas into the CINGSA lity.	т
30.6	Filing	g and Reporting Requirements	
	a)	Seller shall provide Customers with Fuel Use reports each year. This requirement may be met by copy of the report filed pursuant to Subsection 30.6b).	т
	b)	On or before the 15 th of May, Seller will file a report for RCA approval showing actual quantities retained and Fuel Use incurred for the previous Contract Year, along with the calculation of the Injection/Withdrawal Fuel Use Percentage for the ensuing twelve-month period beginning the first day in July.	

RCA No. 733	First Re
	Cancell
	Origina

Revision elling nal Sheet No.

Sheet No.

69 69 **RECEIVED**

January 31, 2020

STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

Т

Cook Inlet Natural Gas Storage Alaska, LLC

SECTION 31 OPERATIONAL PURCHASES AND SALES OF GAS

Seller may from time to time purchase or sell Gas on an interruptible basis as necessary to manage system pressure and maintain system integrity. Purchase or sale of Gas for such purposes shall be made on a nondiscriminatory basis.

RCA No. 733 Firs Ca Ori Sheet No.

70 70

Sheet No.

January 31, 2020 STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

RECEIVED

Cook Inlet Natural Gas Storage Alaska, LLC

SECTION 32 OPERATIONAL FLOW ORDER ("OFO")

- 32.1 General: Seller, in its discretion, shall have the right to issue OFOs when in its judgment it is necessary to maintain or restore the operational integrity of Seller's storage system. Notwithstanding the foregoing, Seller will issue an OFO T when it identifies an occurrence of any condition that is likely to lead to a T curtailment of an FSS Customer's deliverability. Seller will not be required to T issue an OFO:
 - a) To redeliver Gas to any Customer that has not tendered Equivalent Quantities of Gas to Seller's Storage system; or
 - b) To any other pipeline in order to obtain access to quantities of Gas, except to the extent that such quantities of Gas are being transported by such pipeline for the account of a Customer.
- 32.2 Forms of OFOs: An OFO may:
 - a) Direct any Customer to decrease quantities injected or withdrawn or increase/decrease pressures at a Point of Injection/Withdrawal, in accordance with Section 32.4 below; or
 - b) Implement verbal arrangements with transporters; or
 - c) Enable Seller to take or require any other actions as may be deemed necessary by Seller in its judgment in order to maintain the operational integrity of Seller's Storage system.
- 32.3 OFO Operations Conditions: OFOs may be issued in any of the following circumstances:
 - a) To alleviate conditions that threaten the operational integrity of Seller's storage system; or
 - b) To maintain minimum necessary pressures for storage operations.

Т

RCA No. 733	First Revision	Sheet No.	71	RECEIVED
	Cancelling Original	Sheet No.	71	January 31, 2020
	Original	Offeet NO.		STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

- c) The OFO will remain in effect until the operational condition requiring its issuance has been remedied or no longer exists.
- 32.4 Condition to Firm Storage Service Agreements:
 - a) Seller may issue an effective OFO to any Customer under Rate Schedule FSS to (i) decrease quantities tendered to Seller for storage at Customer's Point(s) of Injection/Withdrawal and (ii) decrease quantities requested to be withdrawn from storage at the Point(s) of Injection/Withdrawal, up to Customer's MDWQ. Customer will be required to comply with such OFO within twenty-four (24) Hours of notice.
 - b) Seller will require, as a condition in each FSS Agreement, that Customer guarantee that Seller will have the right to issue an effective OFO to any transporter or owner of Gas being supplied to any Customer, whether or not such OFO is required to be effectuated at the Point(s) of C Injection/Withdrawal. Seller shall issue an OFO pursuant to this Section 32.4b) only if (i) Customer has elected to grant such authority to Seller for the issuance of all OFOs; or (ii) Seller is required to issue the OFO without providing the twenty-four (24) hours notice referenced in Section 32.4a) above.
- 32.5 OFO Notice, Contents and Procedures: Seller shall issue an OFO as expeditiously as is reasonably practicable in the circumstances. Seller shall post and provide Customers with updated information concerning the status of operational variables related to the OFO as soon as it is available. Each OFO will contain the following provisions:
 - a) Time and date of issuance;
 - b) Time that the OFO is considered to be effective (if no time is specified, the OFO shall be effective immediately);

Т

Sheet No. 72

RECEIVED

FEB 1 6 2012

STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

Cook Inlet Natural Gas Storage Alaska, LLC

Cancelling

Original

- c) Duration of the OFO (if none is specified, the OFO will be effective until further notice);
- d) The party or parties receiving the OFO;
- e) The quantity of Gas required to remedy the operational condition requiring the issuance of the OFO; and
- f) Any other terms Seller may reasonably require to ensure the effectiveness of the OFO.
- 32.6 Failure to Comply with OFO: If Customer or its agent fails to comply with the terms of an OFO, for any reason other than Force Majeure on an upstream or downstream pipeline, such Customer shall be: (i) liable for any damages incurred by Seller or any other affected party as a result of such failure and (ii) subject to a penalty of twenty-five dollars (\$25) for each Mcf of Gas that does not comply with such OFO. Notwithstanding anything to the contrary in this Section 32.6, if Customer is required to make a nomination pursuant to an OFO, unless critical circumstances dictate otherwise, no damages and/or penalties will be assessed unless Customer is given the opportunity to correct the circumstances giving rise to the OFO.
- 32.7 Seller's Liability for OFOs. Seller shall not be liable to any person for any costs, damages or other liability associated with the issuance of, or the failure to issue, any OFOs; provided, however, Seller shall be liable for acts of gross negligence or undue discrimination; such standards to be judged in light of the emergency conditions under which OFOs are issued.

TA 4-733

By:

Effective

APRIL 9, 2012

Issued By: Cook Inlet Natural Gas Storage Alaska, LLC,

M. Collen Starring Title: Vice President

M. Colleen Starfing

RCA No. 733

First Revision Cancelling Original Sheet No.

73 73

Sheet No.

January 31, 2020 state of alaska

RECEIVED

STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

Cook Inlet Natural Gas Storage Alaska, LLC

SECTION 33 RIGHT OF FIRST REFUSAL (ROFR)

In the event the term of the FSS Agreement is 12 months or greater and Customer has paid the rates under Rate Schedule FSS, then the Customer's storage capacity under Т the FSS Agreement is subject to the ROFR. In order to initiate the ROFR process, Customer must notify Seller at least nine (9) months prior to the termination date of the FSS Agreement. Upon notification Seller will conduct a thirty (30) Day open season to solicit bids for the Customer's storage capacity. Seller will post information on its website to commence the open season. Posted information will include available volumes, injection and withdrawal rates, term of service, deadline for bids, and other general terms and conditions necessary to be eligible for service. Posted information will describe the process for submitting bids, including a timeline that allows no less than fifteen working days to submit a bid. Within ten (10) Days of the end of the open season Seller will present the Customer the highest bid received in the open season in the form of an FSS Agreement. In order to retain the capacity Customer must return the executed FSS Agreement to Seller within fifteen (15) Days. Seller reserves the right to not accept any bid that is less than the rates under Rate Schedule FSS. In the event no bids are Т received, Seller and Customer may negotiate an extension of Customer's FSS Agreement.

RCA No. 733 Second Revision Cancelling First Revision Sheet No.

74 74

Sheet No.

January 31, 2020 STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

RECEIVED

Cook Inlet Natural Gas Storage Alaska, LLC

SECTION 34 GENERAL TERMS AND CONDITIONS

- 34.1 Volumetric Pricing of Gas. In the event that it either is required or becomes standard practice in the Cook Inlet to price Gas using the heating value of such Gas (i.e., on a MMBtu basis) as opposed to a volumetric basis (i.e., on a Mcf basis), the Parties agree that the Gas under this Agreement will be priced on a MMBtu basis, and will enter into necessary revisions and amendments to this Agreement in order to effectuate such conversion. Unless and until such conversion occurs, for pricing purposes, it shall be assumed that each Mcf of Gas contains one (1) MMBtu.
- 34.2 If Seller fails, because of scheduled or unscheduled interruption of service (other than a Force Majeure Event), to tender for redelivery or accept for storage injection at the Point(s) of Injection/Withdrawal, all or a portion of Customer's С firm injection or withdrawal quantity as the case may be, and if such event extends for a period in excess of fifteen (15) consecutive Days, Customer's subsequent monthly bill shall be reduced by an amount equal to the product of the unfulfilled firm withdrawal quantity times the Reservation rate or the unfulfilled firm injection quantity times the Capacity rate, as the case may be. If Seller fails, because of a Force Majeure Event, to tender for redelivery or accept for storage injection at the Point(s) of Injection/Withdrawal, all or a portion of C Customer's firm injection or withdrawal quantity, and such event extends for more than sixty (60) consecutive Days, Customer's subsequent monthly bill shall be reduced by an amount equal to the product of the unfulfilled firm withdrawal quantity times the Reservation rate or the unfulfilled injection quantity times the Capacity rate, as the case may be.

RCA No. 733	Eleventh Revision	Sheet No.	75	RECEI	VED
	Cancelling Tenth Revision	Sheet No.	75	JUN 29	2023
				STATE OF A REGULATORY COMMI	
	atural Gas Storage				
SECTION 35	RATE SCHEDULES AND	D OTHER CHARGES			
35.1 RA	ATE SCHEDULE FSS - FIF	RM SERVICE			
As	provided for in Section 4.6	6.			
a)	Statement of FSS Storag	e Service Rates:			
				Tariff Rate per Mcf	
	Reservation Rate –				
	ed to Customer's ract Withdrawal Quantity	Monthly		\$4.9712	R
	Capacity Rate – Applied stomer's Total Contract				
Quar		Monthly		\$0.0678	R
FSS	Injection/Withdrawal				
	modity Rate			\$0.0483	I
Over	run Storage Service Rate			\$0.2118	R
Exce Char	ss Storage Service ge:			\$0.0678	R
b)	Subject to the Injection/W 35.3.a.	/ithdrawal Fuel Use P	ercentage	as set out Sectio	n
c)	Billings may be subject t Section 35.4.b., Well Main and local sales taxes.				
d)	Volumetric Pricing of Gas In the event that it either Cook Inlet to price Gas us basis) as opposed to a vo this Tariff will be priced or occurs, for pricing purpo contains one (1) MMBtu.	r is required or becor sing the heating value o plumetric basis (i.e., o n a MMBtu basis. Unle	of such Gas n a Mcf bas ess and unf	s (i.e., on a MMBti sis) the Gas unde il such conversio	น r า

RCA No. 733	Eleventh Revision Cancelling Tenth Revision	Sheet No. Sheet No.	76 76	JUN STATE	EIVED 29 2023 OF ALASKA MMISSION OF ALASKA
Cook Inlet N	atural Gas Storage Al	aska, LLC			
35.2 R	ATE SCHEDULE ISS - INTER	RRUPTIBLE SERV	/ICE		
A	s provided for in Section 5.3.				
a	Statement of ISS Storage S	ervice Rates:			
			Г	Fariff Rate per Mcf	
ISS Rate	9			\$0.2313	R
-	ction/Withdrawal dity Rate			\$0.0483	I
b)	Subject to the Injection/Wi Section 35.3.a.	ithdrawal Fuel Us	e Percenta	ige as set out	in
c)	Billings are subject to the Re 35.4.b, Well Maintenance S also be subject to local sale	urcharges as provi			
d)	Volumetric Pricing of Gas: In the event that it either is Cook Inlet to price Gas using basis) as opposed to a volu this Tariff will be priced on a occurs, for pricing purpose contains one (1) MMBtu.	g the heating value metric basis (i.e., c MMBtu basis. Unl	of such Gas on a Mcf bas ess and unt	s (i.e., on a MMI sis) the Gas un il such convers	Btu der ion

RECEIVED RCA No. 733 14th Revision Sheet No. 77 Cancelling MAY 15 2023 13th Revision Sheet No. 77 STATE OF ALASKA **REGULATORY COMMISSION OF ALASKA** Cook Inlet Natural Gas Storage, Alaska, LLC 35.3 FUEL USE CHARGE As provided for in Section 30.5. Injection/Withdrawal Fuel Use Percentage: a. The Injection/Withdrawal Fuel Use Percentage is the effective percentage assessed on all injections as set forth below: Effective Period <u>Rate</u> April 1, 2012 to May 1, 2013 1.5% May 1, 2013 to April 30, 2014 0.5% May 1, 2014 to June 30, 2015 1.1% July 1, 2015 to June 30, 2016 1.1% July 1, 2016 to June 30, 2017 2.0% July 1, 2017 to June 30, 2018 1.1% July 1, 2018 to June 30, 2019 1.1% July 1, 2019 to June 30, 2020 1.5% July 1, 2020 to June 30, 2021 1.8% July 1, 2021 to June 30, 2022 1.3% July 1, 2022 to June 30, 2023 1.3% July 1, 2023 to June 30, 2024 С 1.3%

July 1, 2023

						RECEIV	ED
RCA No.	733	14th Canceling	Revision	Sheet No. 78	_	JUN 28 20	23
		13th	Revision	Sheet No. 78	_	STATE OF ALAS REGULATORY COMMISSIO	
COO	K INLET NATU	RAL GAS STOR	AGE ALASKA, LLC				7
	35.4 - REGUL	ATORY COST (CHARGE				
	a)			special surcharge applic Company's share of the			
	b)	Regulatory Co	st Charge	3.314% of stc	brage service billing	I	
	Pursuant to U	-23-009(2)		Effective	July 1, 2023		-
			al Gas Storage Alas				J
	By:	Voira K. Smith		Title	e: Vice President	and General Counsel	_

RCA No. 733 Original

Cancelling

Sheet No. 79

Sheet No.

RECEIVED

FEB 1 6 2012

STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

N		OTHER CHARGES	35.5 - SCHEDULE OF
2011	41		
Ľ	\$25.00 for each Mcf	Fee	OFO Penalty F
	of Gas that does not comply with the OFO.	ed for in Section 32.6 to comply with an OFO.	As provide for failure to
		a.	
	12). 1		
		moved from Sheet 76.	L - OFO penalty Fee has been

M. Colleen Starring

RCA No. 733 First Revision Cancelling Original Sheet No. 80

Sheet No. 80

RECEIVED

OCT 0 4 2013

STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

	<u> Sas Storage Alaska, I</u>	LLC		
35.6 – GAS	S STORAGE FACILI	TY TAX CREDIT		N L
facility FSS C Tax Cr	tax credit ("Tax Cred ustomers. CINGSA	it") from the State of will derive no direct dit, and any interes	a \$15,000,000 gas storage Alaska for the benefit of its or indirect benefit from the st accrued thereon, will be	L
ope	ration (as defined in npany must refund a	AS 31.053.032) be	Facility ceases commercial efore January 1, 2022, the f the Tax Credit back to the	
U-10 com	0-51(9), the Company	y has an uncondition nancial benefit the Co	cepted by the RCA in Order al obligation to disburse any ompany receives as a result	
interest Credit Custom disburs	-bearing account. T and any interest ea ers. On an annual b	he Company will act arned for the bene basis through the end hers the amount of th	y deposited it in a separate t as a custodian of the Tax fit of the Company's FSS d of 2021, the Company will ne Tax Credit, not subject to below:	
6	the CINGSA Fact amount of the Tax be one-tenth of the be combined with	ility has continued of Credit to be disburse total Tax Credit (\$1 any interest earned mount to be disburse	re January 1, 2022 in which commercial operations, the ed for that calendar year will ,500,000). That amount will during the calendar year to sed to the FSS Customers	
L - Schedule of S	pecial Contracts (Sec	tion 36) has been mo	oved to Sheet 82.	
			November 18, 2013	

Cancelling

Sheet No.

RECEIVED

OCT 0 4 2013

STATE OF ALASKA **REGULATORY COMMISSION OF ALASKA**

Cook Inlet Natural Gas Storage Alaska, LLC

1 If the CINGSA Facility ceases commercial operations during a N calendar year, there is no Tax Credit to be disbursed for that calendar year and the remaining Tax Credit will be refunded to the State in accordance with AS 43.20.046(h) and (i). The Total Annual Disbursement for that calendar year, which will also be the final Total Annual Disbursement, will include all interest earned. and not yet distributed, up to the date of disbursement.

2. If the CINGSA Facility continues commercial operations through January 1, 2022, the Total Annual Disbursement for calendar year 2021 will contain the final one tenth of the Tax Credit and will be the final Total Annual Disbursement. It will also include all interest earned, and not yet distributed, up to the date of disbursement.

- b) The Company will calculate each FSS Customer's pro rata share of the Total Annual Disbursement based upon the Customer's average monthly contracted MSQ for the calendar year as a percentage of the average monthly total of all FSS Customers' contracted MSQ for the calendar year to which the Total Annual Disbursement applies.
- c) For the Total Annual Disbursement related to calendar year 2012, the Company will apply each FSS Customer's pro rata share of the Total Annual Disbursement as a credit to the first billing for storage service that occurs at least 30 days following RCA approval of this Section 35.6.
- d) For subsequent calendar years, including the final Total Annual Disbursement, the Company will calculate and apply each FSS Customer's pro rata share of the Total Annual Disbursement as a credit to the billing for storage service for January of the following year. For example, the Company will apply each FSS Customer's pro rata share of the Total Annual Disbursement calculated using the 2013 calendar year information on the billing for January 2014 storage services.

TA 12-733

Effective November 18, 2013

Issued By: Cook Inlet Natural Gas Storage Alaska, LLC M. Collen Starring By:

Title: President

M. Colleen Starring

RCA No. 733	Second Revision Cancelling	Sheet No.	82
	First Revision	Sheet No.	82

SECTION 36	SCHEDULE OF SPEC	CIAL CONTRACTS		
<u>Customer</u>		Contracted Service	Contract Effective Date	
ENSTAR Nat	ural Gas Company	Firm Service	October 17, 2011	
Chugach Elec	ctric Association, Inc.	Firm Service	October 17, 2011	
Alaska Electri Cooperative,	0,	Firm Service	October 17, 2011	D

D – Deleted Anchorage Municipal Light & Power special contract

RCA No. 733	Third Revision	Sheet No.	83	RECEIVED
	Cancelling			NOV 30 2021
	Second Revision	Sheet No.	83	STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

SECTION 37	CREDITING OF NON-FIRM REVENUES TO FSS CUSTOMERS	
37.1	Pursuant to Order U-18-043(15), this Section 37 is effective for Non-Firm revenues billed on or after August 16, 2019.	C C
37.2	The Company will share 100% of certain Non-Firm revenues collected with its FSS Customers on a monthly basis.	
	a) The FSS revenues eligible to be shared will be revenues charged under the "Overrun Storage Service Rate" and the "Excess Storage Service Charge" set out in Section 35.1 of this Tariff. The ISS revenues eligible to be shared will be revenues charged under the "ISS Rate" and the "ISS Injection/Withdrawal Commodity Rate" set out in Section 35.2 of this Tariff and any related interest charged on unpaid balances. This will begin with the revenues billed in the first month following the effective date of this Section 37 ("Eligible Non-Firm Revenue").	
	b) Related Regulatory Cost Charges, and sales taxes will not be shared.	
	c) Only Eligible Non-Firm Revenue that has been collected from Customers will be shared.	
37.3	At the beginning of each month, prior to the billing of its FSS Customers, the Company will identify the amount of Eligible Non-Firm Revenue that was collected in the previous Month (if any).	
	a) The collected Eligible Non-Firm Revenues will be the amount to be shared with the FSS Customers ("Shared Amount").	с с
	b) The Shared Amount for the Month will be allocated to the FSS Customers on a pro rata basis measured by contracted Maximum Storage Quantity ("MSQ") and Contract Withdrawal Quantity ("CWQ"), with a 50% weight on each of the capacity reservation factors. The contracted MSQ and CWQs are set out in the Appendix A to each FSS Agreement.	с с с с

RECEIVED
AUG 2 9 2019
STATE OF ALASKA TORY COMMISSION OF ALASKA
Ν
e Non- date of llowing e FSS
ay)
SQ and apacity
ers' ed nt 19 51 50 50 50
sharing annual

RCA No. 733

Original Cancelling

Sheet No.

Sheet No.

85

85

RECEIVED OCT 3 1 2019

Cook Inlet Natural Gas Storage Alaska, LLC

STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

TA 35-733	Effective	: December 16, 2019		
	And proceeding of the second and the			
	f) The term "Surcharge" means a Well Maint under this Section 38.	enance surcharge established		
	e) The term "Qualified Well Maintenance V Maintenance Work Package that utilizes a	-		
	d) The term "Eligible Well Maintenances Co Qualified Well Maintenance Work Packag in Section 38.3 below.			
	c) The term "Drill Rig" means the machine onshore operations, the rig includes virt quarters. Major components of the rig inc pumps, the derrick or mast, the drawwork the drillstring, the power generation equip A workover rig is a type of Drill Rig.	ually everything except living clude the mud tanks, the mud s, the rotary table or topdrive,		
	b) The term "Coiled Tubing Unit" means the p to run a coiled tubing operation. Four bas the coiled tubing reel to store and transpo- injector head to provide the tractive effort tubing string, the control cabin from w controls and monitors the operation, and t the necessary hydraulic and pneumatic components. The dimensions and capac components determine the size and leng can be used on the unit. Pressure-control the equipment to provide the necessary during normal operating conditions and co emergency control.	sic components are required: ort the coiled tubing string, the to run and retrieve the coiled hich the equipment operator the power pack that generates power required by the other sities of the coiled tubing unit of coiled tubing string that equipment is incorporated into control of well pressure fluid		
	a) The term "AOGCC" shall mean the Alas Commission.			
38.2	Definitions: In addition to the definitions set out in Section 2 below shall have the following meanings for the p	•		
38.1	38.1 Purpose: This surcharge mechanism is established to allow the Company to recover actual, prudently-incurred Well Maintenance expenses associated with work that requires a Drill Rig.			
SECTION 38	WELL MAINTENANCE SURCHARGE MECH	ANISM	Ν	

RCA No. 733	Original Cancelling	Sheet No. Sheet No.	86 86		EIVED
				STATE C)F ALASKA
Cook Inlet N	atural Gas Storage A	laska, LLC	í - 77	REGULATORY COM	MISSION OF ALA
g)	The term "Well Mainter necessary to maintain towards, its previous primarily mitigates or re geophysical risks, and sidetracks.	or to return an exi fully functional c mediates engineeri	sting well to apacity. We ng rather thar	or substantially Il Maintenance n geological and	N
h)	The term "Well Mainten an individual well design remediation purpose. I 20 AAC 25.280 require 403) by the AOGCC. A include multiple Sundr engineering costs, mob other related costs neo integrity of the well so th	ned to achieve a sta t may include indivi an Application for A specific Well Main y Approvals and ilization costs, dem cessary to restore	ted Well Mair dual work ph Sundry Appro- tenance Wor includes the includes the includes the function a	ntenance or well hases that under ovals (Form 10- rk Package may materials, pre- osts, testing and il or mechanical	
i)	The term "Wireline" mea single-strand or multist wells. Slickline and e placement and retrieva and valves, as well as equipment.	rand wire or cable e-line are types of I of wellbore hardv	for interventi Wireline use vare, such as	ion in oil or gas ed for selective s plugs, gauges	
38.3 El	igible Well Maintenance Cos	ts			
a)	Total Eligible Well Ma Surcharge shall be sp Determination of Surch Section 38.	ecified in the tariff	advice filing	supporting the	
b)	Eligible Well Maintena prudently-incurred cost		clude only a	ctual, historical,	
c)	Eligible Well Maintenar Qualified Well Maintena done with wireline ar Maintenance Work Pa Package does not utiliz	ance Work Packag nd/or electric line ckage, but not if t	e. It includes tools as pa he Well Mai	s costs for work art of the Well intenance Work	
d)	Eligible Well Maintena personnel.	ance Costs shall	exclude cos	sts of CINGSA	
e)	Eligible Well Maintenan	ce Costs shall not r	eflect carryin	ig costs.	
TA 35-733				r 16, 2019	

RCA No. 733	Original	Sheet No.	87	RECEIVED
	Cancelling	Sheet No.	87	DEC 0 6 2019
	Cancelling	Sheet No.	87	

STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

Cook Inlet Natural Gas Storage Alaska, LLC

- 38.4 Separate Surcharges: A separate Surcharge shall be levied for each Qualified N Well Maintenance Work Package.
 - a) Each new or revised Surcharge shall be filed as a separate tariff advice letter, or as part of a general rate case, along with a Determination of Surcharge Rates and supporting documentation for the costs to be recovered.
 - b) Total costs to be recovered through a Surcharge shall be amortized over 3 years (36 months), unless a longer period is required and adequately supported.
 - c) The dollar amount of the Surcharge will be applied to FSS Customers on a pro rata basis measured by the contracted storage quantities and withdrawal capacities of the FSS Customers, with a 50% weight on each capacity reservation factor. In accordance with this, the Surcharge rates to be applied to contracted MSQ and CWQ for FSS Customers will be calculated in a Determination of Surcharge Rates as set out in Section 38.5 below. A Surcharge rate for ISS Customers to be applied to the maximum amount stored each Month will also be calculated in the Determination of Surcharge Rates as set out in Section 38.5 below from the FSS Surcharge rates, following the methodology used in the calculation of the Rate Schedule ISS-ISS Rate set out in Section 35.2.
 - If the total amount of contracted MSQ or CWQ for FSS Customers used d) in a Determination of Surcharge Rates changes during the time a surcharge is in place, the Company may revise the Surcharge by filing a revised Determination of Surcharge Rate that reflects the surcharge amounts already assessed, as well as the new total contracted amounts. The amortization period of the revised surcharge shall be the remaining time left from the original amortization period. The revised Surcharge may be implemented immediately upon filing, subject to subsequent Commission review and approval, and will be effective for all billings subsequent to the revision date.
 - The Surcharge shall stay in place until the total amount of Eligible Well e) Maintenance Costs for the Qualified Well Maintenance Work Package have been recovered. Any over-collection will be refunded to the FSS Customers on a pro rata basis measured by the contracted storage quantities and withdrawal capacities of the FSS Customers, with a 50% weight on each capacity reservation factor.

December 16, 2019

RCA No. 733	Original Cancelling	Sheet No. Sheet No.	88 88	RECEIVED OCT 3 1 2019
Cook Inlet N	latural Gas Storage A	laska, LLC		STATE OF ALASKA REGULATORY COMMISSION OF ALASKA
1) Qual 2) Eligii 3) Amo 4) Mon 5) FSS 6) FSS 7) FSS ((line 8) FSS ((line 9) ISS 9 stor 10) Effe 38.6 D A N is to	etermination of Well Maintena lified Well Maintenance Packa ble Well Maintenance Costs rtization Period (Months) thly Amount (line 2 ÷ 3) Total MSQ Contracted Quan Total CWQ Contracted Quan Surcharge Rate-Applied to N e 4 X 50%) ÷ Line 5) Surcharge Rate-Applied to C e 4 X 50%) ÷ Line 6) Surcharge Rate-Applied to ma ed for the Month Per Mcf ((line ective for billings on or after etermination of Well Maintena ssume that the Eligible W laintenance Package ("Examp 36 months, the total FSS MS otal FSS CWQ Contracted Qu ovember 1, 2019. The comp e as follows:	age ID tities (Mcf) ntities (Mcf) ISQ Monthly Per Mcf WQ Monthly Per Mcf aximum quantity of ga e 8 ÷ 30.4) + Line 7) ance Surcharge Rate /ell Maintenance Co ple") totaled \$1,000,0 SQ Contracted Quan uantities is 150,000 M	f f as osts for a 000, the amo tities is 11,00 Mcf and the e	ntization period 00,000 Mcf, the effective date is
1) Quali 2) Eligib 3) Amor 4) Montl 5) FSS 6) FSS 7) FSS ((line 8) FSS ((line 9) ISS Si store	fied Well Maintenance Packag le Well Maintenance Packag le Well Maintenance Costs tization Period (Months) hly Amount (line 2 ÷ 3) Total MSQ Contracted Quanti Total CWQ Contracted Quanti Surcharge Rate-Applied to MS 4 X 50%) ÷ Line 5) Surcharge Rate-Applied to CM 4 X 50%) ÷ Line 6) urcharge Rate-Applied to max d for the Month Per Mcf ((line ctive for billings on or after	ge ID ities (Mcf) ities (Mcf) SQ Monthly Per Mcf NQ Monthly Per Mcf ximum quantity of gas	\$ 11,0 1 \$ \$ \$ \$	9 000,000 36 27,778 000,000 50,000 50,0013 50.0926 50.0043 1, 2019

RCA No. 733	Original Cancelling	Sheet No.	89	RECEIVED
	5	Sheet No.	89	OCT 3 1 2019

STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

- 38.7 Separate Listing on Bills: The total amount billed for each Surcharge will be **N** separately listed on each billing.
- 38.8 Annual Accounting: The Company shall file an annual accounting of the collections under each effective Surcharge at the same time it files its annual report to the Commission.

CA No. 733	Fourth Revision Cancelling Third Revision	Sheet No. Sheet No.	90 90	RECEIV FEB 27 20 STATE OF ALA	D23 ska
ook Inlet N	Natural Gas Storage A	laska IIC		REGULATORY COMMISSI	ON OF ALA
	Effective Well Maintenance Su	urcharges:			
а	i) Filed as TA36-733				
1) Qu	alified Well Maintenance Pac	kage ID		CLUS-3 Workover	
2) As	of the billing for January 20 amount authorized for recover and the charge has expired.			Workover	C C C
	SS Surcharge Rate-Applied t	o CWQ Monthly Per M	/lcf	\$0.0000	С
4) ISS	ne 4 X 50%) ÷ Line 6) S Surcharge Rate-Applied to r		as	\$0.0000	с
	ored for the Month Per Mcf ((li ctive for billings on or after	ne & ÷ 30.4) + Line 7)		February 10, 2023	С
b) Filed as TA49-733				
1) Qu	alified Well Maintenance Pac	kage ID		CLUS-5 Workover	
2) Elig	gible Well Maintenance Costs	i		\$ 3,340,625	
	nortization Period (Months)			36	
,	nthly Amount (line 2 ÷ 3) S Total MSQ Contracted Qua	ntition (Mof)		\$	
,	S Total CWQ Contracted Qua			150,000	
7) FS	S Surcharge Rate-Applied to ne 4 X 50%) ÷ Line 5)	, , , , , , , , , , , , , , , , , , ,		\$0.0042	
8) ÈS	SS Surcharge Rate-Applied t	o CWQ Monthly Per N	/lcf	\$0.3093	
9) ISS	ne 4 X 50%) ÷ Line 6) S Surcharge Rate-Applied to r pred for the Month Per Mcf ((li		as	\$0.0144	
	ective for billings on or after	100.00.4 · LINC /)		November 1, 2022	
с	:) This section is current	ly not in use.		-	
A 51-733		Effective: A	pril 1	3, 2023	

Effective: August 5, 2020

Cook Inlet Natural Gas Storage Alaska, LLC

Reserved for Future Use

RCA No. 733	Original
	Cancelling

Sheet No.

Sheet No.

91

91

RECEIVED

January 31, 2020

STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

Ν

Reserved for Future Use

Original

Cancelling

92

Sheet No.

Sheet No.

STATE OF ALASKA

Ν

RCA No. 733

92

January 31, 2020

RECEIVED

REGULATORY COMMISSION OF ALASKA

Sheet No.

93

93

January 31, 2020

RECEIVED

STATE OF ALASKA **REGULATORY COMMISSION OF ALASKA**

Ν

Cook Inlet Natural Gas Storage Alaska, LLC

Reserved for Future Use

RCA No. 733	Original
	Cancelling

Effective: August 5, 2020

Pursuant to U-19-025(20)
Issued By: Cook Inlet Natural Gas Storage Alaska, LLC

SECTION 39 CUSTOMER COMPLAINTS

Customer complaints may be registered at the offices of the Company at 3000 N Spenard Road, Anchorage, Alaska in person or by telephone during the regular N business hours set out in Section 1.2 or by mail. Such complaints will be promptly N handled on an individual basis and appropriate steps will be taken to resolve the situation. In the event the complaint is not resolved to the Customer's satisfaction, the Customer may notify the Regulatory Commission of Alaska.

Ν

Sheet No.

Sheet No.

94

94

RCA No. 733

Original Cancelling

> STATE OF ALASKA REGULATORY COMMISSION OF ALASKA



January 31, 2020

RCA	No.	733	0
1.07.1	110.	100	<u> </u>

Original Cancelling Sheet No.

95

95

RECEIVED

Sheet No.

STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

Ν

Cook Inlet Natural Gas Storage Alaska, LLC

SECTION 40 FORMULA RATE MECHANISM

- 40.1 Application: This Formula Rate Mechanism ("FRM") provides for an annual revision ("FRMAR") to the Company's Rate Schedules FSS and ISS (Sections 35.1 and 35.2, respectively). Rate calculations and adjustments required by this Section 40 shall be determined on a Facility-Wide revenue requirement basis.
 - A. No provision contained within this Section 40 will limit the Company's ability to file a General Rate Change Application or limit the RCA's authority over rates.
 - B. Except as provided in Section 40.1.C, the Company will file a FRMAR by tariff advice letter on or before June 15 (or on the next Business Day after June 15 if June 15 is not a Business Day) of each year.
 - C. The Company is exempt from filing a FRMAR for any Test Year for which:
 - 1. the Company files a General Rate Change Application,
 - 2. the Company is directed by the Commission to file the information required by 3 AAC 48.275(a) thus initiating a General Rate Case for a given Test Year, or
 - 3. a General Rate Change Application or General Rate Case is pending before the Commission.

40.2 Definitions:

In addition to the definitions set out in Section 2 of this Tariff, the terms listed below shall have the following meanings for the purpose of this Section 40:

- A. The term "3 AAC" means the regulations of Commission as set out in Title 3 of the Alaska Administrative Code. The numbers following "3 AAC" are references to a specific regulation section.
- B. The term "AOR" means the Company's annual operations report filed with the Commission as required by AS 42.05.451(b). It includes the FERC Form No. 2.
- C. The term "AS" means the Alaska statutes. The numbers following "AS" are references to a specific statute section.
- D. The term "Facility-Wide" means including all aspects of the CINGSA Facility used to provide FSS and ISS.

RCA No. 733	First Revision	Sheet No.	96	RECEIVED
	Cancelling	Cheet Ne	00	JUN 02 2021
	Original	Sheet No.	96	STATE OF ALASKA
				REGULATORY COMMISSION OF ALASKA

Г

E.	The term "FERC" shall mean the Federal Energy Regulatory Commission.	
F.	The term "Filing Date" means the date the Company files a FRMAR.	
G.	The term "Final Order" means the last substantive order in a General Rate Case approving permanent rates, or accepting a settlement setting permanent rates, that becomes final and is not subject to further reconsideration by the RCA or appeal.	
H.	The term "FRM" shall mean formula rate mechanism under this Section 40.	
I.	The term "FRMAR" means the FMR annual revision under this Section 40.	
J.	The term "General Rate Case" means a docketed proceeding before the RCA to review the entire revenue requirement (including cost of capital) and cost of service of the Company and involves the RCA review of the information required by 3 AAC 48.275(a). A General Rate Case may be initiated by a General Rate Change Application filed by the Company or by the action of RCA. A "Concluded General Rate Case" is one where a Final Order has been issued.	
K.	The term "General Rate Change Application" means a request to change the rates of the Company's FSS and ISS Customers filed in accordance with AS 42.05.411 and 3 AAC 48.275.	
L.	The term "Test Year" means the twelve Months ending December 31 of the preceding calendar year.	
М.	The Term "TYEB" means the Test Year end balance, i.e., the balance at the end of the Test Year.	
N.	The terms "Uniform System of Accounts" and "USoA" mean the Uniform System of Accounts for Class A natural gas companies prescribed by Part 201 of the FERC regulations (18 C.F.R. Part 201) and required for use by 3 AAC 48.277(5). The number following "USoA" is a specific USoA account number.	

July 19, 2021

RCA No. 733		riginal ancelling	Sheet No.	97	FEB 22 2021
		U	Sheet No.	97	STATE OF ALASKA REGULATORY COMMISSION OF ALASKA
Cook Inlet	Natu	ral Gas Storage	e Alaska, LLC		
40.3	FRMA	R Calculation			Ν
	("F be sc	१R") that will adjust low. The Company s hedules showing the	ate an annual, Facility applicable rate schedu hall request recovery of e computation of any a hall be calculated accor	ules prospe f its total RF adjustments	ctively as set out R and shall include to the Test Year
		RR	R = OM + DEP + OT + R	RI + IT	
	W	here:			
	ma an tre	aintenance expenses d measurable chan	curred, reasonable ar s incurred during the T ges and prepared con r accepted in the Comp	est Year a sistent with	djusted for known the rate making
	1.	shall also be limite Filing Date and tha in which the rate	rable adjustments shal d to those changes tha t are more than likely to s will be in effect a recedent regarding	at have occ o continue t nd are co	curred prior to the hrough the period
			n the level of salary a st Year, or are knowr		
		b. There will be n and general co	o pro forma adjustmen sts.	ts to alloca	ted administrative
	2.	each year based o Year, but the metho	nd overhead allocation n the latest component odology used will be tha ncluded General Rate (factors use t approved	ed during the Test

RCA No. 733

Original

97

RECEIVED

RCA No. 733 First F Cance Origin

Г

First Revision Cancelling Original

Sheet No.

98

98

March 13, 2023

Sheet No.

STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

٦

RECEIVED

JAN 27 2023

Cook Inlet Natural Gas Storage Alaska, LLC

C.	DEP = depreciation and amortization expense calculated on end-of Test Year plant and asset balances at the depreciation and amortization rates approved in Docket U-18-043 and TA50-733 or at the rates approved or C accepted in a subsequent Concluded General Rate Case.
D.	OT = taxes other than income tax from the Test Year prepared consistent with the rate making treatments approved or accepted in the Company's last Concluded General Rate Case.
E.	RI = return on prudently incurred investment calculated as the Company's rate of return (weighted average cost of capital) multiplied by the Test Year rate base.
	 Rate of return shall be the weighted average cost of capital approved or accepted in the Company's last Concluded General Rate Case, except as provided in (a).
	a. If the Company refinances existing long-term debt or issues new or additional long-term debt, it will recalculate its weighted average cost of capital using its approved or accepted return on equity, its approved or accepted capital structure, and its new cost of debt. The new cost of debt calculation will include the costs of issuing the debt and any gain or loss on retiring the old long- term debt including any retirement or refinancing premium.
	2. Rate base is prepared in accordance with the following:
	a. Known and measurable adjustments shall be limited to (f) and (j) below and also be limited to those changes that have occurred prior to the Filing Date and are consistent with the ratemaking treatments approved or accepted in the Company's last Concluded General Rate Case.
	b. Gas utility plant shall be the TYEB of USoA accounts 101-105.
	 Accumulated depreciation reserve shall be the TYEB of USoA accounts 108 & 111.

RCA No. 733	Origi Canc	inal celling	Sheet No.	99	RECEIVED FEB 22 2021
	oune	Jennig	Sheet No.	99	STATE OF ALASKA REGULATORY COMMISSION OF ALASKA
Cook Inlet N	atural	Gas Storage	Alaska, LLC		
	d.	Plant completed 106.	not classified shall be	the TYEB	of USoA account
	e.	Gas stored-base	e gas shall be the TYEB	of USoA a	account 117.1.
	f.	lead/lag time pe	ng capital allowance sl eriod (whether positive e Company's last acce	e or nega	tive) accepted or
	g.	Materials and su 156.	pplies shall be the TYEI	B of USoA	accounts 154 and
	h.	Prepayments sh	all be the TYEB of USo	A account	165.
	i.	USoA account 1	ets shall be the TYEB 82.3 that were approve ne Company's last Conc	d or accep	oted to be included
	j.	calculated for re in rate base (e) excess accumula such ADIT amou credits, and if the to rate base. The	ferred income taxes ("/ gulatory purposes, dire kcluding Cash Working ated deferred income ta ints are debits, they sha e net amount is a debit ne amount included in the e TYEB for the Test Yea	ctly assoc Capital) axes ("Exc Il be netted , then it sh the rate ba	ciated with an item and shall include cess ADIT"). If any d against the ADIT hall be an addition
		rate that oc measurable,	nacted changes in the curred during the Tes shall be reflected in th e (per paragraph F-2 be	t Year, o ne calculat	r are known and

Sh	eet	No.	
JII	CCL	INU.	

100

100

RECEIVED FEB 22 2021

Sheet No.

STATE OF ALASKA **REGULATORY COMMISSION OF ALASKA**

Cook Inlet Natural Gas Storage Alaska, LLC

Original

Cancelling

k. Net asset retirement obligation shall be those accounts for asset retirement obligations ("ARO") excluding any amounts and accounts included in the gas utility plant accounts (USoA accounts 101-105) or accumulated depreciation and amortization reserves (USoA accounts 108 and 111). If any such ARO amounts are debits, they shall be netted against the ARO credits, and if the net amount is a debit, then it shall be an addition to rate base. The amount included in the rate base calculation for net ARO shall be the TYEB.	N
 Customer advances for construction shall be the TYEB of USoA account 252. 	
F. IT = income tax for the adjusted Test Year, adjusted for known and measurable changes occurring after the Test Year and before the Filing Date, and prepared consistent with the rate making treatments approved or accepted in the Company's last Concluded General Rate Case.	
1. Income tax expense shall be calculated as follows:	
 Equity Return shall be calculated as rate base multiplied by the weighted cost of common equity from the Company's last Concluded General Rate Case, 	
 After-tax net return shall be calculated as the equity return, plus the annual amortization of allowance for equity funds used during construction and capitalized into plant, plus or minus the net annual amortization of protected and unprotected Excess ADIT, as approved or accepted in Docket U-18-043, 	
 c. Income tax expense before amortization of Excess ADIT shall be calculated as the after-tax net return multiplied by a composite gross-up state/federal income tax factor calculated as ((1/(1 – incremental state corporate income tax rate)) X (1/(1 - incremental federal corporate income tax rate))) – 1, and 	
 Income tax expense shall be calculated as the income tax expense before amortization of Excess ADIT, plus or minus the net annual amortization of Excess ADIT. 	
2. Statutorily enacted income tax rate changes that occurred during the Test Year, or are known and measurable, shall be annualized, per the income tax expense calculation described in 1-c above. The Company shall comprehensively account for, including establishing a regulatory liability or asset to account for, any such change in income tax expense in the calculation to ensure recovery of income tax expense under new and old income tax rates.	

Effective:

Issued By: Cook Inlet Natural Gas Storage Alaska, LLC

101

101

RECEIVED FEB 22 2021

Sheet No.

STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

Ν

Cook Inlet Natural Gas Storage Alaska, LLC

Original

Cancelling

(1	Cost Allocation: Once the annual RR has been determined as provided in A-F above, it shall be allocated between fixed costs and variable costs using the same cost classification ratios that were approved or accepted in the Company's last Concluded General Rate Case.	
		 For example, if the cost of service allocation approved or accepted in the Company's last Concluded General Rate Case allocated 99% of the RR to fixed cost and 1% to variable cost, then the RR for the FRMAR calculated in a-f above would also be allocated 99% to fixed costs and 1% to variable costs. 	
ł	 (Derivation of FSS Storage Service Rates: FSS Storage Service Rates shall be calculated consistent with the methodology approved or accepted in the Company's last Concluded General Rate Case. The billing units used will be the approved FSS Agreement contract quantities (CWQ and MSQ) in effect during the period in which the rates will likely be in effect.	
		1. FSS Reservation Rate: Unless it is calculated differently in the Company's last Concluded General Rate Case, the FSS Reservation Rate shall be calculated by taking 50% of the RR fixed costs calculated in G. above, divided by the total FSS CWQ in effect for the period for all approved FSS Agreements, divided by 12 ((RR fixed costs X 50%)/total FSS CWQ/12 months).	
		 FSS Capacity Rate: Unless it is calculated differently in the Company's last Concluded General Rate Case, the FSS Capacity Rate shall be calculated by taking 50% of the RR fixed costs calculated in G. above, divided by the total FSS MSQ in effect for the period for all approved FSS Agreements, divided by 12, divided by 1,000,000 ((RR fixed costs X 50%)/total FSS MSQ/12 months/1,000,000). 	

Sheet	No	
Sileel	INU.	

102

102

RECEIVED FEB 22 2021

Sheet No.

STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

Cook Inlet Natural Gas Storage Alaska, LLC

Original

Cancelling

		N
3	8. FSS Injection/Withdrawal Commodity Rate: Unless it is calculated differently in the Company's last Concluded General Rate Case, the FSS Injection/Withdrawal Commodity Rate shall be calculated by taking 100% of the RR variable costs calculated in G. above, divided by the total amount of all FSS injections and withdrawals for the Test Year.	
4	A. Overrun Storage Service Rate: Unless it is calculated differently in the Company's last Concluded General Rate Case, the FSS Overrun Storage Service Rate shall be calculated by taking the FSS Reservation Rate calculated in 1. above, divided by 30.4 (average number of days in a month), then adding the FSS Injection/Withdrawal Commodity Rate calculated in 3. above.	
5	5. Excess Storage Service Charge: Unless it is calculated differently in the Company's last Concluded General Rate Case, the Excess Storage Service Charge shall be the FSS Capacity Rate calculated in 2. above.	
b	Derivation of ISS Storage Service Rates: ISS Storage Service Rates shall be calculated consistent with the methodology approved or accepted in the Company's last Concluded General Rate Case.	
1	. ISS Rate: Unless it is calculated differently in the Company's last Concluded General Rate Case, the ISS Rate shall be calculated by taking the FSS Reservation Rate calculated in H.1. above, divided by 30.4 (average number of days in a month), then adding the FSS Capacity Rate calculated in H.3. above.	
2	2. ISS Injection/Withdrawal Commodity Rate: Unless it is calculated differently in the Company's last Concluded General Rate Case, the ISS Injection/Withdrawal Commodity Rate shall be the FSS Injection/Withdrawal Commodity Rate calculated in H.3. above.	

CA No. 73	3	Original Cancelling	Sheet No.	103	RECEIVED FEB 22 2021	
			Sheet No.	103	STATE OF ALASKA REGULATORY COMMISSION O	F ALASKA
ook Inle	t Na	tural Gas Storage A	laska, LLC			
40.4		MAR Schedules addition to the tariff advice le	etter, the FRMAR fili	ing shall inc	N lude the following:	
	A.	Schedules of Test Year N Rate Base, Weighted Cos Requirement in similar forn schedules and its Derivation	t of Capital, and No nat to the first four pa	ormalized To ages of the	est Year Revenue Company's 275(a)	
	В.	A schedule and explanation known and measurable ch		ı, annualizir	ng, pro forma, and	
	C.	A schedule and explanation	on of all amortized ex	xpenses.		
	D.	A schedule in a format sim delineates total operating categories from pages 31 year balances, previous ye explanation for each accor percent from the previous	revenue and the 7-325 of the FERC ear balances, and pe ount that increased	relevant ec Form 2 th ercent varia	uivalent expense at include current nce, as well as an	
	E.	A schedule that includes total administrative and ge to ENSTAR'S construct CINGSA, and reimbursabl	eneral expenses and ion, Alaska Pipeli	d then the a	amounts allocated	
	F.	A schedule showing the (CWQ and MSQ) in effect quantities in effect during t	t during the test per	riod and, if	different, contract	
			ne period in which u			

104

104

Sheet No.

STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

Ν

FEB 22 2021

RECEIVED

Cook Inlet Natural Gas Storage Alaska, LLC

Original

Cancelling

- 40.5 FRMAR Evaluation and Review Procedures:
 - A. A copy of the FRMAR filing will be provided to the Commission Staff, the FSS Customers, and the Office of the Attorney General, Regulatory Affairs & Public Advocacy Section ("RAPA") at the time it is filed with the Commission, along with any Excel worksheets with working formulas used to create the schedules, exhibits or attachments in the filing. The Company will schedule regular and recurring meetings at least two times per year with the FSS customers and RAPA, including a meeting before CINGSA's filing of its FRMAR to discuss any issues that may be relevant to the FRMAR. The meetings may be incorporated into the biannual meetings committed to in the Second Stipulation accepted by the Commission in Order U-19-025(19).
 - B. The Commission Staff shall review and evaluate the FRMAR filing, and may request clarification and additional data, and the Company shall provide the same.
 - C. The Company shall work in good faith to promptly and fulsomely answer all questions raised by the Commission Staff. If the Company and the Commission Staff agree that any calculations or schedules in the FRMAR filing should be revised, the Company shall file with the Commission the resulting adjusted rate calculations, revised tariff sheet, or revised FRMAR schedules.

RCA	No.	733	
-----	-----	-----	--

Original Cancelling Sheet No.

105

105



Sheet No.

STATE OF ALASKA REGULATORY COMMISSION OF ALASKA

Ν

Cook Inlet Natural Gas Storage Alaska, LLC

40.6 FRMAR Effective Date and Further Proceedings A. A filed FRMAR becomes permanent at the end of the notice period described in AS 42.05.411 unless the Commission suspends the filing in accordance with AS 42.05.421. B. If the Commission suspends the filing, the Commission may allow the filing to take effect on an interim basis, subject to refund. 1. If the Commission suspends a FRMAR filing, the Company shall have the option to supplement its filing and request, and convert the filing to a General Rate Case Application. a. The Company shall notify the Commission that it will supplement its filing and request, and convert the FRMAR filing to a General Rate Case Application, within 30 days of the Order suspending the FRMAR filing. b. After giving the notification, the Company shall have an additional 45 days to supplement its filing and convert it to a General Rate Case Application. c. The FRMAR filing may be updated to reflect those changes that have occurred prior to the date on which the Company supplements its filing under this Section 40.6.B, including updating adjustments made in the FRMAR filing. Any new or revised known and measurable adjustments shall be limited to those changes that have occurred prior to the date the Company supplements its filing. d. The Company shall have the option to request a revision to any interim rates upon its filing converting the FRMAR to a General Rate Case Application.